(Occasional Papers)

Financial stability considerations on bail-in

by Maurizio Trapanese



Questioni di Economia e Finanza

(Occasional Papers)

Financial stability considerations on bail-in

by Maurizio Trapanese

The series Occasional Papers presents studies and documents on issues pertaining to the institutional tasks of the Bank of Italy and the Eurosystem. The Occasional Papers appear alongside the Working Papers series which are specifically aimed at providing original contributions to economic research.

The Occasional Papers include studies conducted within the Bank of Italy, sometimes in cooperation with the Eurosystem or other institutions. The views expressed in the studies are those of the authors and do not involve the responsibility of the institutions to which they belong.

The series is available online at www.bancaditalia.it.

ISSN 1972-6643 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

FINANCIAL STABILITY CONSIDERATIONS ON BAIL-IN

by Maurizio Trapanese*

Abstract

This paper provides an economic analysis of the challenges to the bail-in regime in terms of its ability to affect the structure of incentives within large banks and to solve the problem posed by intermediaries that are classed as 'too big to fail' (TBTF). It first explains the peculiar role of large banks in the modern financial system and the impact of their failures on financial stability – both of which aspects are invoked to justify their differential treatment in a crisis. This discussion of the rationale for a bail-in regime helps understand the economic basis and the minimum conditions for the tool to be applied efficiently. The paper shows that European bail-in regulations involve complex rules and mechanisms that can be accompanied by discretionary aspects, potentially leading to uncertainties in the application of the procedure and difficulties in in achieving its objectives. A credible regulatory framework and clear decision-making processes would strengthen market discipline and reduce moral hazard. Finally, the paper argues that the bail-in procedure seems to be less viable, from a functional perspective, in the instance of a banking crisis with potentially systemic implications.

JEL Classification: G01, G21, G28, H12.

Keywords: financial crises, banks, financial policy and regulation, crisis management.

DOI: 10.32057/0.QEF.2025.968

^{*} Bank of Italy, DG Economics, Statistics and Research.

1. The objective of this paper (introduction) ¹

This paper undertakes an economic analysis of the challenges facing a bail-in regime to enhance market discipline, reduce moral hazard and end the too-big-to-fail (TBTF) problem.² To this end, the paper examines the main theoretical contributions of the economic literature and some aspects of the EU primary legislation to develop financial stability considerations on the appropriateness of the bail-in tool to pursue its original objectives.

The focus is on large banks, given their peculiar role in the modern financial systems and the impact of their failures on financial stability. ³ The paper examines the minimum conditions needed for an efficient application of bail-in, and for which the rationale to mandate the involvement of the private sector in a banking crisis is justified. The paper also draws some examples from the EU resolution framework, ⁴ where the bail-in might not be fully effective in enhancing market discipline and reducing moral hazard.

¹I wish to thank Alessio De Vincenzo, Emilia Bonaccorsi Di Patti, Antonio Di Cesare, Sebastiano Laviola, Luigi Federico Signorini, and Valerio Vacca, for their very useful comments and observations on earlier versions of this paper, and Gwyneth Schaefer for the language review. The opinions expressed here do not necessarily represent the positions of the Bank of Italy. Any errors remain the author's own responsibility.

² The TBTF problem is an economic theory and an approach in banking supervision, which states that certain systemically important financial institutions are so large and interconnected with the rest of the financial system that their distress or disorderly failure would cause significant disruption to the wider financial system and the economic activity. In this respect, the systemic costs of their failures are much larger than the costs of any individual bail-out. Consequently, under the TBTF approach, these institutions have to be supported by the government, when facing serious troubles. For a review of the literature on causes, consequences, and costs of public bailouts, and the main elements of policy framework aimed at addressing the TBTF problem, see: Otker-Robe et al. (2011); Strahan (2013); Tucker (2013); and Dewatripont (2014).

³ In this paper, 'large banks' are intended as those banks whose size and risk profile are deemed to be of such importance that their failure would trigger a wider financial crisis and threaten the real economy. This category may include either the global systemically important banks (G-SIBs) defined by the Financial Stability Board (FSB) and the Basel Committee on Banking Supervision (BCBS), or the banks qualified as 'significant' within the EU Banking Union (BU).

⁴ As to the EU resolution framework, this paper refers exclusively to the EU primary legislation, i.e., the Bank Recovery and Resolution Directive (BRRD), and the Single Resolution Mechanism Regulation (SRMR). In doing so, it does not include the second tier regulation (technical standards, guidelines, technical advices, recommendations, etc.) developed by the EU technical authorities in the field of resolution and crisis management.

The starting point of the analysis is the 'regime change' that took place after the global financial crisis (GFC) of 2007-08, when public authorities (governments, central banks, and supervisors) were called on to end the TBTF problem within the post-GFC regulatory repair, recognizing that public bail-outs of large banks with taxpayer money would no longer be warranted. This 'regime-change' occurred because the GFC bail-outs of large banks, with immense recourse to tax-payer money, proved to create moral hazard, impair market discipline, and produce a destabilizing impact on public finance. Accordingly, during the post-GFC years new resolution regimes have been established, in the majority of the world's developed economies. The new regimes aim to limit the systemic consequences of large bank failure, while safeguarding the sustainability of public finances.

The introduction of the bail-in tool is at the centre of these new resolution regimes. Bail-in involves replacing the implicit public guarantees on large banks balance sheets, with which fractional reserve banking systems have been operating for decades, with a system of private penalties. In the case of a banking crisis, it mandates the bank's shareholders and some creditors to absorb the relative losses, through a pre-defined waterfall, and to support the burden of recapitalization (Gordon and Ringe, 2015; and Avgouleas and Goodhart, 2015). This is because the bail-in tool could replace bankruptcy proceedings, allowing the bank to continue as a going concern, thus avoiding the disruptive effects of a liquidation or break-up of an ailing bank. In doing so, it is supposed to eliminate the TBTF subsidy for large banks.

Some economists have considered the bail-in as the modern alternative to the two traditional tools used to address banking crises famously described by Walter Bagehot in his influential book Lombard Street: 1) provide central bank liquidity to illiquid (but solvent) banks; or 2) wind down insolvent ones. Bail-in is a 'third way', in the sense that it seeks to self-insure banks, so that a rescue with public money becomes unnecessary (Ringe, 2016). In order to restore market discipline and reduce moral hazard, bail-in compels banks to internalize the costs of the risks they assume. This also could reduce the amount of losses in a bank failure, especially where the bail-in

⁵ The first pillars of this policy strategy are in G-20 Leaders Statement of 2009 (G-20, 2009), which has been translated in a regulatory framework by the FSB, through a building-block approach consisting of a high number of reports/documents/recommendations spanning from 2010 to 2019. For a selected overview, see: FSB (2010a); FSB (2010b); FSB (2011); FSB (2013); FSB (2014); FSB (2015); and FSB (2019).

regime allows for earlier intervention and closure than a bail-out mechanism (Avgouleas and Goodhart, 2015).

Against this background, this paper contends that bail-in represents an important development, since it improves the overall policy mix available to authorities in a crisis. Ex post, it has proven to be a tool that authorities tend to manage with care, and this explains why in the post-GFC regulatory reforms a significant layer of supervisory discretion has been included to complement the automaticity of the rules.

The post-GFC reforms take into account past historical experience, which shows that the competent authorities have constantly used significant discretion when managing and resolving banking crises, with a record of positive outcomes on average. However, this paper argues that if these degrees of discretion are framed in a complicated regulatory framework, and where the decision-making structure is based upon the interaction of multiple authorities, the cumulative effects of these factors may create severe uncertainties.⁶

The economic literature agrees that market discipline can be enhanced and moral hazard can be reduced if investors foresee the price of their capital instruments and anticipate the expected losses in case they are bailed-in with a reasonable degree of certainty. Accordingly, investors would price a bank's equity capital based upon only the default probability embedded in the bank's assets and not with a view to distorting implicit public guarantees. If these conditions fail, the original objectives of the bail-in may be impaired. In this respect, the paper draws some examples from the EU resolution framework where this may apply. From a functional perspective, the paper explains why the bail-in approach may be less viable in a crisis with potential systemic implications.

⁶ The relationship between regulatory complexity and systemic risk in the main regulatory, prudential, and resolution frameworks of the last two decades is examined in Trapanese (2022). In this analysis, the resulting outcome in terms of excessive regulatory complexity might turn out to be costly, and sub-optimal for crisis prevention. Since modern finance is characterised by uncertainty (rather than risk), the paper advocates a regulation rebalanced towards simplicity, which may produce Pareto-improving solutions, and encourage better decision making by authorities and regulated entities.

⁷ As to the EU resolution framework, this paper refers to the EU primary legislation, i.e., the Bank Recovery and Resolution Directive (BRRD), and the Single Resolution Mechanism Regulation (SRMR). In doing so, it does not include in the analysis the second tier regulation (technical standards, guidelines, technical advices, recommendations, etc.) developed by the EU technical authorities in the field of resolution and crisis management.

This paper is organised as follows. Section 2 explains the key features emphasizing the specificity of large banks, in particular those from their liabilities side and corporate structure. Section 3 describes the main economic models used to understand the destabilizing effects of large bank failures. Section 4 examines the policy rationale underpinning the involvement of the private sector in banking crises. Section 5 explains the economic mechanism of the bail-in tool and its effects on banks' behaviour. Section 6 elaborates the approaches set forth in the economic literature regarding the minimum conditions for an efficient bail-in regime. Section 7 draws some examples from the EU resolution framework where such conditions may be challenged. Section 8 offers some reflections on the capacity of the bail-in to end the TBTF problem. Section 9 concludes.

2. Why large banks are different

Large banks are subject to a differentiated treatment in the case of a crisis, because they perform a role in the intermediation process very different from that of the other banks or financial institutions, and because of the existence of negative externalities associated with their failure. Their balance sheets and business models exhibit unique features, in terms of a higher degree of interconnectedness with the rest of the financial system, and a more complex corporate structure, often associated with a significant cross-border reach.

Most importantly, large banks have 'financial liabilities', which carry on specific functions, whose amount is far greater than other liabilities. These 'financial liabilities' include a large array of products, such as foreign and domestic (uninsured) deposits, checking accounts, commercial papers, trading derivative liabilities, insurance policies, and repurchase agreements (repos). These liabilities are more than claims to a future stream of income; some of them (e.g., bank deposits or repos) are a source of liquidity; others (e.g., derivatives or insurance policies) are also risk-shifting instruments.

The specificities of large banks' have been extensively modelled in the economic literature. Sommer (2014), and McAndrews at al. (2014) explain in detail why a financial firm's insolvency

⁸ This concept of financial liabilities excludes from the scope of the analysis leasing, factoring, lending or mortgage companies. These firms do not issue financial liabilities, but raise their funds from banks and on the financial markets. In this respect, these firms are not financial firms for our purposes. For these considerations see: Sommer (2014); and McAndrews at al. (2014).

is different in nature from the insolvency of an ordinary (commercial or industrial) firm. In their models, large banks perform a special role given the amount and the functions of the 'financial liabilities' held in their balance sheet, which explain the economic reasons for a special insolvency treatment of large banks.

The objective of optimal insolvency laws would be to protect these 'financial liabilities', in order to preserve their liquidity or risk-shifting functions, which can have systemic implications. Generally, this objective is obtained at the cost of impairing the other liabilities. This can happen because this cost is less than the cost of impairing 'financial liabilities', which have a bearing on liquidity or risk shifting functions, and can be enormous, if it takes the form of systemic risk. For these reasons, 'financial liabilities' are given priority treatment in insolvency law, i.e. they have to be paid first in line, and often first in time, before payments to other creditors are made. Some 'financial liabilities', like bank deposits or insurance policies, can be firm-specific. This means that they are more valuable if kept with the firm's business, than paid off in an insolvency distribution. This is exactly why in the case of a large bank, a reorganization is more efficient than a liquidation, because a reorganization preserves the value of the firm-specific liabilities (Sommer, 2014).

Large banks are not single entities and usually they have an international scale for their business and operations. There are several reasons for this corporate structure: the different regulatory and prudential frameworks governing the various segments of financial intermediation (e.g. banking and insurance); the fiscal policies of the firm, and the specificities of the business model undertaken. The parent company is typically a bank or holding company. Most of the affiliates are centrally controlled, and share management, personnel, business, reputation, and operations. ⁹ The result is a complex organizational structure, which can be decomposed in an insolvency procedure with difficulty.

For these reasons, the classical resolution mechanisms, such as those applied to small or medium-size, primarily domestic, banks with relatively simple organizational structures and funding sources may be not equally effective in the crisis of a large bank. To complicate things further, the viability of a bank is to be ensured on a continuous time basis, since its operations cannot be suspended for an indefinite period. Things are even more complicated, given that large

⁹ Some of the subsidiaries have 'financial liabilities', others do not. Banks, insurers, securities dealers, derivatives dealers are in the first sample; mortgage banks, venture capital firms, asset holding companies are in the second.

banks crises give very little warning. In order to preserve liquidity and solvency at the system level, the management of these crises must be decided in a very short time span.

In the case of smaller banks, authorities can exercise statutory powers to break-up the bank, by transferring assets (e.g., performing loans) and liabilities (e.g., retail and other short-term deposits) to a 'good bank' (the 'bridge bank'), in such a way that the 'bridge bank' remains solvent and can be sold at a later stage. The remainder is left in the original bank (the so-called 'bad bank'). Those creditors that are transferred to the 'good' bank are preferred to the other creditors. When a large bank goes into troubles, the division of assets and liabilities to be transferred into 'good' or bad' entities is extremely complex. Since these banks generally take the form of complex and international groups, with a significant share of assets and liabilities governed by foreign laws or held through foreign branches or subsidiaries, the effectiveness of nationally based resolution tools may be in question.¹⁰

It is often the case that large banks have little cross-border branching, since their preferred option involves the establishment of subsidiaries, which act as legally separate entities in foreign countries. This tends to complicate the cross-border application of insolvency proceedings even further. In the case of a large cross-border banking group crisis, cooperation is more difficult, since jurisdictions tend not to cede their primacy on their local entities, being potentially conducive to sub-optimal outcomes.¹¹

¹⁰ The insolvency procedure of international firms is far more complex than that of domestic firms. Cross-border insolvency may require multiple and competing insolvency administrations of a firm, with each administrator using its own law to conduct the proceedings on the entity falling under its legal responsibility. The norm is that of a central administrator with other jurisdictions in a supporting role, conducting 'ancillary proceedings' that assist the main proceeding. In a liquidation, the ancillary proceeding collects assets and distributes them to a central receiver for distribution. In a reorganization, the ancillary jurisdiction enforces the stay and does whatever asset collections are necessary. For a detailed examination of the interaction among the central and the ancillary proceedings in a cross-border banking insolvency, see Sommer (2014), and McAndrews at al. (2014), who base their analysis mainly on the US model.

¹¹ However, there are some incentives for cooperation. In the case of ordinary firms, local liquidations will destroy value, and the automatic stay will buy time for cooperation; industrial insolvencies are common and large industrial firms are spread over the world. This supports reciprocity: a jurisdiction may agree to a subordinate role now in return for a central role in the future. These incentives are far weaker in financial firm insolvencies, since they are rare and asset specificity less intense. See: Sommer (2014); and McAndrews at al. (2014).

3. Large bank failures and financial stability

3.1 The "monetary view" of banking crises

Large banks are highly interconnected with the rest of the financial system. The failure of a large bank can trigger social costs through a variety of channels, on a scale that is much greater than in the case of a smaller bank. A large bank failure has the potential to impose losses across the financial system and the economy at large that are a multiple of the losses incurred by that bank's investors. ¹² The high degree of maturity transformation and liquidity mismatch incurred by these banks exacerbates their vulnerability to a sudden decline in the value of their liquid assets or a sudden increase in demand for liquidity. Moreover, their assets tend to become illiquid in times of general stress, thus aggravating the shock and potentially spreading contagion across the board.

The economic literature has provided consistent models to shape the destabilizing effects of a large bank's failure on the financial system. Some of them underline that banks are special since they issue deposits or other money-like liabilities. ¹³ This monetary view of banking crises has been around since it was first explained by Friedman and Schwartz, in their famous book on the monetary history of the US, who argued that the Great Depression was aggravated by the failures of banks that contracted the supply of bank liabilities (that is money).

The production of money-like instruments and/or services can be considered as one of the most important functions exercised by banks. The demand deposits issued by banks are the main factor explaining the failures of banks in the case of depositors' run. The failures of large, complex and interconnected banks are not only costly in terms of destruction of asset values, but also destabilizing, as these failures have the potential to transfer risks to other financial institutions,

¹² As an example, the market capitalization of Lehman Brothers at the end of January 2007 was about \$60 billion, and that of the troubled US banks during the GFC was approximately \$1.2 trillion. The economic consequences of the GFC on the US economy for the years 2008-09 was much larger. The US suffered fiscal outlays of about \$5 trillion, and the US economy contracted by 3.5 per cent in 2009 (about \$9 trillion). These US figures do not include of course the costs incurred in the rest of the world economies. For these data, see Armour (2014).

¹³ A bank is not an ordinary commercial company, which can continue its business while in insolvency. A bank cannot, since no other entity would voluntarily deal with an insolvent bank, neither for a very short period. Simply, the essence of bank's specificity is its solvency; by definition, an insolvent bank is not a going concern. As to the classical references for the bank's specificity, see: Baltensperger (1980); Diamond and Dybvig (1983); Fama (1985); Goodhart (1987); and Bernanke and Blinder (1988).

determining more instability for the system as a whole. ¹⁴ These liabilities confer money-like or liquidity services that may be impaired in a failure and are runnable. If the holders of these 'financial liabilities' run in order to avoid the consequences of a possible failure, the bank has to borrow to replace the funding that has been lost, and/or sell assets quickly. The asset sales can lead to deeply discounted prices (that is, fire sales), and to losses, further undermining the solvency of the bank, and spread distress on other banks or financial firms with similar portfolios of assets.

McAndrewes at al. (2014) show some evidence in support of this role of uninsured 'financial liabilities' in banking crises. Using data on all failed banks resolved by the Federal Deposit Insurance Corporation (FDIC) from 1985 to 2011, first, they say that banks more reliant on uninsured 'financial liabilities' in the year before failure, experience larger contractions in these liabilities in the ensuing year. Second, they show that the cost of failures to the FDIC was increasing in the amount of uninsured liabilities on a bank's balance sheet in the year before the crisis. Finally, they take these findings as conclusive evidence that a greater reliance on uninsured liabilities leads to runs and fire sales, which in turn increase the costs of the failure. These findings are consistent with the hypothesis that the holders of these liabilities are prone to runs.

The 2023 US banks' failures have drawn attention to the potentially destabilizing role of uninsured deposits in modern banking systems, given the increased speed of withdrawals allowed by technological progress, and experienced during the crises. The 2023 events revealed the special vulnerability of banks to runs by small and medium-size firms that rely on uninsured bank deposits to meet their business and other high-frequency operational needs. Like other runnable liabilities, they risk creating contagion across the financial system. If the proper functioning of the financial system is seriously impaired, the real economy receives less liquidity to fund productive activities. The lack of credit drives down total output, with potentially destabilizing effects on economic growth and social welfare. As we know, these are the classical arguments in favour of bank capital regulation and supervision.

3.2 Large banks and "tail risks"

Other strands of the economic literature consider banks as intermediaries that gather savings of households and lend to productive firms, emphasising their role in delegated monitoring and

¹⁴ The failure of a small bank is costly, in terms of lost local output, but it does not threaten the smooth function of the overall financial system. Aschraft (2005) explains how even small bank failures destroy the private information that banks develop about their clients so that borrowers become credit constrained after the bank's failure.

information sharing among different economic sectors.¹⁵ In this context, contagion within the financial system is particularly harmful, since these banks collectively perform functions that are essential for the functioning of the real economy. They not only make credit available to non-financial firms, but they also perform valuable screening and monitoring functions in relation to funded business projects (Bernanke, 1983; and Armour, 2014).

Typically, large banks have a high degree of leverage, implying that an increasing level of risk can be embedded in the their balance sheets. ¹⁶ Given the dimensions of their balance sheets, large banks can be exposed to the so-called 'tail risks', ¹⁷ which is difficult to measure, given that even a small increase is hard to verify, and banks have creditors unable to monitor their conditions on a continuous time basis. Diversification and hedge schemes may not suffice in reducing the assets' variance. The materialization of these risks has potential destabilizing effects on the individual banks and the system as a whole. ¹⁸ In the case of large banks, the spread of contagion occurs in an amplified manner, since these banks are highly interconnected, meaning that problems at one can easily be transmitted to others, via a number of channels. ¹⁹ The mechanisms for the

¹⁵ The classical references for the theory of asymmetric information, delegated monitoring, and bank specificity, are: Leland and Pyle (1977); and Diamond (1984).

¹⁶ For a comprehensive survey comparing the degrees of leverage across firms, banks and countries, also within an historical perspective, see Kalemli-Ozcan et al. (2012). According to these authors, debt to equity ratios of 1:1 are usual for ordinary firms; financial firms' ratios are much higher, about 15:1–30:1 for banks and securities firms, and somewhat less for insurers.

¹⁷ Tail risk is the financial risk of an asset or portfolio of assets moving more than three standard deviations from its current price, above the risk of a normal distribution. Tail risks include low-probability events arising at both ends of a normal distribution curve. However, financial markets are not perfect as they are largely shaped by unpredictable human behaviour, so that the distribution of returns is not normal, but skewed. Tail risk materializes when a rare, unpredictable, and very important event occurs, resulting in significant fluctuations in the value of the stock. These tail events are often referred to as black swan events, whose likelihood is greater than the one predicted by traditional strategies.

¹⁸ According to Sommer (2014), in the period 1980-2010 at least seven episodes can be viewed as tail events. These are the following: equity market crash of 1987; 1994 bond market crisis; 1997 Asian financial crisis; 1998 Russian financial crisis and the Long-Term Capital Management episode; dot-com bubble collapse; subprime mortgage crisis, the crisis of the Lehman Brothers, or the collapse of the AIG group during the GFC.

¹⁹ A large bank can be a counterparty for a high number of other market participants. The liabilities of one bank can be the assets of others, which can be devalued in the case of a financial stress: A bank can be forced to fire-sell assets leading to depressed prices for assets that other institutions use as collateral. A large bank's default or even the fear of a possible default determines panic that spreads to other institutions. Brunnermayer (2009), Acharya et al. (2011),

transmission of the crisis to other parts of the financial system are of different nature since large banks are highly interconnected each other.²⁰

Systemic risk events are not common, and often trace back to multiple causes, but their preconditions hinge in the balance sheet structure of large banks. In particular, the key reason underlying such systemic consequences can be traced to the large banks' liability structure, which – as shown in the previous section—implies heavy reliance on uninsured 'financial liabilities'. This view about the contribution of uninsured financial liabilities to systemic instability does not exclude that also illiquid asset holdings or organizational complexity coupled with a multi-layered and internationally based corporate structure have the potential to produce similar effects. More precisely, in terms of the behaviour of large and systemic banks, these features are normally additive, and as such may produce multiple and self-reinforcing effects while banks approach failure.

The non-bank financial intermediation sector has created innovative forms of liabilities, such as repos, that provides services functionally equivalent to traditional banks' deposits. ²¹ Similarly, other forms of uninsured financial liabilities, such as commercial paper issued by banks, are also demandable at par for large customers that request the financial firm to buy back its paper. As a result, a large amount of the funding of big financial firms is made up of uninsured financial liabilities, which provide money-like services, and create the conditions for customers' runs.

Troger (2018), and Gorton and Metyrick (2012) offer a detailed description of the different types of systemic consequences from a failure of a large bank as occurred during the GFC.

²⁰ According to Sommer (2014), the mechanism for the transmission of the instability is not so important; only the results matter. It could be a pure panic attack, with bad news for one bank imputed to all; or a markdown of an asset class by one bank triggers markdowns for all banks; Or an industry-wide hedging model goes broken; or a large bank's liabilities are another large bank's assets; Or a clearinghouse goes in difficulty blocking liquidity. A role can be played by leverage and liquidity stress. In this case, the asset side also becomes illiquid, since asset liquidity dries up precisely when a large bank most needs this liquidity.

²¹ According to Gorton and Metrick (2012), repos can be considered a type of money, because they are liquid, functionally demandable at par due to their largely overnight tenor, and able to function as an overnight store of value.

4. The policy rationale for a bail-in regime

4.1 Public bail-outs and their effects on the financial systems

Government decisions involving public-sector bail-outs have been considered for decades a rational and policy-based response to avoid the potentially destabilising effects of a large bank's failure, averting fire sales and contagion. Since banks are not ordinary commercial firms, governments have often decided to recapitalize them using taxpayer funds, thereby preventing further damages from a bank failure. However, the use of taxpayer money may undermine financial stability if it gives rise to a situation of unsustainable public finances.²²

The policy rationale for governments' bail-outs and a critical evaluation of their effects on the financial system, have been extensively examined in the economic literature. ²³ It is stressed that the recourse to public-sector bail-outs allows market participants to predict this kind of decisions, so as to include an implicit government's guarantee in the price of bank capital, with the effect to reduce large banks' default probability. Such implicitly guaranteed banks, considered too-big-to-fail, show lower risk premiums and are able to raise capital at lower costs.

A number of empirical studies confirms these distorting market prices mechanisms.²⁴ In this context, bank equity holders and managers have the incentives for an excessive risk-taking behaviour (moral hazard), which ultimately leads to an increased level of leverage, and to

²² In some cases, there can be positive (non-risk adjusted) returns from bail-outs on government equity stakes. For example, this has been the case of JP Morgan Chase, Wells Fargo, Goldman Sachs, Crédit Agricole, BNP Paribas, and Société Generale, for the years 2008 and 2009. For this evidence, see Hertig (2012).

²³ For essential references, see: Zhou et al. (2012); Avgouleas and Goodhart (2014); and Troger (2018). The main economic arguments to replace public bailouts with private sector in banking crises have been put forward even before the GFC. See, for example, Eichengreen and Ruehl (2000).

²⁴ As to the US banking markets, Schweikhard and Tsesmelidakis ((2012) show that in the US CDS markets the US major banks have enjoyed risk premiums on their debt significantly different from the rest of the US banking system in the years of the GFC. Santos (2014) indicates that in the 20 years preceding the GFC, investors systematically accepted much lower spreads on bonds issued by the largest US banks as compared to bonds issued by smaller banks. Tsesmelidakis and Merton (2012) estimate the funding advantage of the 74 largest US banks benefiting from implicit government guarantees. Ueda and Weder-Di Mauro (2012) find an average increase of 70 basis points in the structural subsidy as measured in the credit ratings of large banks in the years 2007, 2008, and 2009. Morgan and Stiroh (2005) show that the relationship between bond spreads and ratings was lower for TBTF banks than for other financial intermediaries during the 1990s.

inefficient investment decisions.²⁵ Furthermore, market discipline is hampered when a bank's risk bearing capacity is not the guiding criterion for its capital market pricing.

The implicit government bail-out guarantee to larger banks could also have an undesirable effect on the competitive conditions of banking markets. The advantage of lower funding costs could give larger banks the possibility to cut margins ad to eliminate smaller competitors. In this way, the implicit subsidy distorts the pattern of the intermediation towards larger banks and away from smaller banks and non-banks (Gropp et al., 2011; and Afonso et al., 2014). However, this shifted intermediation also could have positive outcomes, since larger banks are normally considered to be more regulated, safer and less risky. Moreover, much of the subsidy could create better conditions for the banks' borrowers, mainly in the form of lower interest rates (Kovner at al., (2014).

Adrian and Ashcraft (2012) have generalized to the case of the implicit government guarantees the conclusions reached by Merton (1977) and Merton and Bodie (1993), who have been the first to have modelled the effects on market discipline of an explicit public guarantee, that is the banking deposit insurance being established as part of the safety net. The same holds for the level playing field for the provision of financial services to the economy, which is undermined in the case bail-outs make bank's funding costs depending on the financial strength of their sovereign.

After the GFC, public authorities recognised that public bail-outs of large banks could worsen moral hazard, and impair market discipline. In order to limit the systemic effects stemming from a large bank's failure, and safeguard the sustainability of public finance, they mandated the definition of a regulatory framework in which the private sector could be requested to bear most of the costs of these failures. These new regulatory regimes have been designated in such a way to undo any government implicit subsidy and credibly connect banks' funding to the level of risks incurred, in order to reduce the excessive risk taking, induced by moral hazard.²⁶

²⁵ As to the effects of implicit subsidies on bank behaviour and risk taking in the international credit markets, see the following papers: Gropp et al. (2011); Gadanetz et al. (2012); Brandao-Marques et al. (2013); and Afonso et al. (2014).

²⁶ An economic analysis of the bail-in as a tool to reduce moral hazard and mitigate systemic risk, can be found in the following: Coffee (2010); Avgouleas and Goodhart (2014); and Gordon and Ringe (2015).

4.2 Bail-in under bounded rationality

The bail-in tool has the objective to ensure the involvement of the private sector in a banking crisis, constraining the risk bearing capacity of the economic agents who invest in bank capital. ²⁷ Its purpose is to provide a mechanism with which to restore balance sheet stability, at the expense of its shareholders and creditors, without the need for publicly funded capital injection. It should not be a penalty for the failure; it primarily aims at an efficient allocation of risk and does not intend to express any assessment of who is to be called for the bank's failure. (Gleeson, 2012; Troger, 2015; and Troger, 2018).

For this tool to work efficiently, the condition is that the private sector involvement should not undermine the viability of the essential functions of the resolved bank. In other words, the losses to be borne by holders of bank capital should not reduce the bank's clients confidence on the capacity of the bank itself to continue to provide its critical services (liquidity provision, and risk-shifting) during resolution and even afterwards. According to Zhou et al. (2012) and Sommer (2014), only under these conditions can bail-in preserve the incentives attributed to insolvency proceedings, but avoid overly disruptive effects, preventing liquidity stress, averting file sales and ultimately the disorderly liquidation of financial contracts. Along the same line of reasoning, Schleifer and Vishny (2011) offer a theoretical discussion about the effects that complex and time-consuming ordinary insolvency proceedings may have on undermining the confidence of the banks' clients and destabilizing the financial markets.

In doing so, bail-in enhances the capacity of the resolution framework to eliminate the incentives a disorderly termination of financial contracts in run-like scenarios that potentially go beyond the bank in trouble (Troger, 2018). ²⁸

²⁷ In the case of a purely domestic bank, the effect of shifting from bail-out to bail-in primarily is to transfer the burden of losses from one set of domestic payers (the tax-payers), to another, the savers. On a theoretical ground, it is far from clear whether and why the latter have broader capacity to absorb bank rescue losses than the former. One argument is that savers or their financial agents have made an ex ante choice to purchase the claim on the bank, performing a monitoring function, whereas the taxpayer had no such option. The counterargument is that individual savers, small- or medium-size pension funds do not have the expertise to act as the economic theory mandates. Thus forcing them to pay for a bank failure would not per se be a welfare improvement. These difficulties could be overcome in the case the regulatory framework defines severe limits on the possibility to acquire banks' liabilities. For these considerations, see Avgouleas and Goodhart (2015).

²⁸ Bail-ins aim to avoid the need for formal insolvency proceedings, by restructuring the bank's balance sheet and ensuring the continued survival of the bank, without immediate dismemberment. To this extent, bail-ins can be

The bail-in tool restores the basic insolvency principle whereby shareholders and creditors must bear the bank's losses, according to a certain order of priority, before any public funds are used (Avgouleas and Goodhart, 2015). In this respect, since the continuity of the critical functions performed by large banks has to be safeguarded, shareholders and creditors cannot suffer losses at the end of the insolvency procedure, where creditors satisfy themselves pro rata on what is left after the liquidation of the debtor's assets. It is necessary that this occurs ex ante.

The contribution of the private sector to bear the costs of a banking crisis has to be designated carefully; otherwise, it can act as a crisis accelerator, since the haircuts imposed on the investors' financial instruments may make it harder for troubling banks to refinance themselves, thus increasing pro-cyclical systemic crisis effects (Avgouleas and Goodhart, 2014). If investors base their decisions on full rationality, bail-in cannot trigger a systemic crisis, since it has the capacity to limit the loss bearing to a sub-set of market participants (Sommer, 2014).

This certainly holds when only one bank, whatever its size, experiences serious difficulties. Even when the number of troubled banks increases, recapitalization from the private sector can be a successful way-out strategy. The only preconditions for such an outcome is that: 1) the banking sector as a whole should have adequate capital and other liabilities subject to bail-in; and 2) the liquidity provided by the central bank can be accessed without limits (Philippon and Salord, 2017; Troger, 2018).

However, the hypothesis of full rationality as a guiding criterion for the decisions of the economic agents is purely theoretical. With the more common hypothesis of bounded rationality, runs may occur even if investors have no reason to believe that their bank is in trouble. Since Diamond and Dybvig (1983), who have studied modern bank runs at deposit-taking institutions, we have understood that the financial system can be shocked in proportions that go well beyond the above-mentioned safeguards.

Drawing from the US experience of the 2007-08 financial crisis, Avgouleas and Goodhart (2014) stressed how the bail-in mechanism may lead to irrational fears that similar haircuts also could be imposed on creditors of other banks, thus inducing investors to run. The spread of investor panic can be mitigated if there is in the system a sufficiently large layer of long-term, non-runnable,

18

considered another kind of resolution tool, which, like temporary public ownership, preserves the bank as a whole as a going concern, imposing losses on creditors and shareholders (Gleeson, 2012).

high quality and easy to be bailed-in liabilities that credibly ensures the boundaries of private sector involvement in a crisis. The objective is that the other creditors should be reassured the haircut tool will not reach their claims as well (Micossi et al., 2014).

Under bounded rationality and with an unclear scope and possible extension of the crisis, only the existence of a second line of defence - in the form of a credible government backstop to be activated for situations in which the bail-in instrument is insufficient - has the potential to arrest panic among market participants. Even when there is a credible private sector involvement, a strong public backstop is required for all runnable contracts on the liability side of the banks' balance sheet; on the contrary, incoming crises could be accelerated.

In modern financial systems, runs also may occur outside the area of deposit-taking credit institutions. Several studies have verified the scope of the public backstops needed to ensure stability also with respect to non-bank financial institutions (Gorton, 2009; and Troger, 2018). In this case, a key issue refers to the question of how far these backstops should reach to ensure the stability of these segments of the financial system too, that is what parts of their balance sheets should be protected. This does not impair the capacity of the private sector's involvement in a crisis to foster market discipline and contain moral hazard and excessive risk-taking behaviour.

5. The economic mechanism of the bail-in

5.1 The prisoner's dilemma and Pareto optimality

Bail-in can be understood as the modern alternative to the two tools traditionally adopted by central banks to manage banking crisis, i.e. to provide central bank liquidity for banks that are illiquid, and to wind down insolvent banks. ²⁹ It represents the "third way" to handle a failing bank, by seeking to *self-insure* banks, so that a rescue with public money becomes unnecessary. A strong intellectual support for this interpretation of the bail-in may be found in the difficulty to distinguish

²⁹ Walter Bagehot, has famously described these two tools in his influential book *Lombard Street*, published in 1873. Bagehot advocated that a lender of last resort in a crisis should lend at a penalty rate to solvent but illiquid banks that have adequate collateral. The doctrine has been criticised as having no place in the modern interbank markets. Bagehot's prescription aims to eliminate the coordination problem of investors at the base of the crisis. It is still a useful guide for action when the interbank market stalls. It makes clear that discount-window lending to entities in need may be necessary in a crisis. Bagehot's doctrine, however, is easy to state and hard to apply. It requires the central bank to distinguish between institutions that are insolvent and those that are merely illiquid. For essential references, see Ciocca and Sannucci (1990); and Ringe (2016).

a liquidity crisis from an insolvency situation. Many economists assert that the dividing line between the two situations may become increasingly blurred, since insolvency and illiquidity are virtually indistinguishable during a real crisis.

Furthermore, the possibility to carry out a comprehensive assessment is usually limited due to time constraints. Since the aim of the bail-in is to preserve the bank as a going concern, its 'trade' creditors (short-term creditors, payment services customers, securities and trading exposures, etc.) must be preserved, so that the bail-in is to apply to the long-term creditors of a bank (bondholders, and holders of subordinated debt). In such a situation, the crucial question becomes how to distribute the emerging losses, and decide which creditors should accept what quantities of these losses.

In the case of the application of the bail-in, bondholders usually have two choices, plus a third one, which may become very inconvenient. Under the first choice, bondholders can decide to agree with the proposed operation, and exchange their subordinated bonds for shares, hoping that the capital increase is successful and that the value of the shares rises in the subsequent periods. The second choice implies that they hold their subordinated bonds until maturity, hoping that a bail-in does not apply. The third, which can be less convenient, is to try to sell their bonds now, accepting the likely losses.

The key aspects of the game theory and the prisoner's dilemma emerge in this difficult decision-making process. ³⁰ Bondholders have every interest in not accepting the proposed exchange, in order to keep their bonds and, upon maturity, have the loan returned. By doing so, however, they can incur in a risk, because, if all bondholders make this choice, the chances of resolution of the bank, which leads to bail-in and cancellation of subordinated bonds, increase. On the other hand, the success of the operation seems complicated, since the majority of bondholders

³⁰ The prisoner's dilemma, one of the most well-known concepts in the modern game theory, is a paradox in decision analysis, in which two individuals acting in their own self-interest do not produce the optimal outcome. It represents a paradigmatic example of how strategic thinking between individuals can lead to suboptimal outcomes for both players. The typical prisoner's dilemma is set-up in such a way that both parties choose to protect themselves at the expense of the other participant. As a result, both participants find themselves in a worse state than if they had cooperated with each other in the decision making process. Solutions to prisoner's dilemmas focus on overcoming individual incentives in favor of the common good. In the real world, most economic and other human interactions are repeated more than once. This allows parties to choose strategies that reward cooperation or punish defection over time. For essential references, see the following: Schneider and Shields (2022); and Stanford Encyclopedia of Philosophy (2024).

would have to put aside their interest, accept risks and losses and convert their bonds into shares. From a theoretical point of view, the main effect from the bail-in in a crisis might be to solve the prisoner's dilemma, and to restore Pareto optimality³¹ to the class of outcomes of individual choices, thus making it easier for authorities to intervene, even in a cross-border context.

This economic substance has been considered effective for ensuring private sector participation in the refinancing of a troubled bank, also in the case of a large bank operating in a number of different national legal frameworks. The primary problem with this model of privately funded recapitalization is that it is more or less impossible to identify every significant creditor of a large and international bank in any reasonable timescale, and even harder to persuade them to agree amongst themselves in the short period available to those charged with resolving a bank. These issues are more acute where a bank is significantly dependent on capital markets funding with a dispersed bondholder group. In this context, the prisoners dilemma will arise, and orchestrating all the parties towards a consensual solution in a weekend timetable may just prove too challenging (Gleeson, 2012).

As the Lehman episode showed, the path towards a consensual solution among all the international actors over a tight timetable has proved to be too challenging. After the GFC, the FSB and the IMF have fostered progress towards international agreements in order to define ex ante coordination mechanisms for the actions and decisions to adopt by the relevant authorities in a crisis. It might contribute to reach coordinated solutions in an international context, with so many actors involved, and within the given time constraints.

The trade-off between ex post and ex ante optimality has been taken into account by a number of economic models, which tend to examine the linkages between bail-in policies, the cost of debt, and the bank managers behavior in dynamic settings (Pandolfi, 2018). In this respect, bail-in is considered ex post the first-best policy for recapitalizing failing banks, but ex ante it can generate a time inconsistency problem, just like bail-outs. In the previous paragraph, we have seen that the main problem with bail-outs is the weakening of market discipline induced by implicit public guarantees. In the case of the bail-in, the problem is the implied increase in banks' funding

³¹ Pareto optimality is an economic state where resources cannot be reallocated to make one individual better off without making at least one individual worse off. It implies that resources are allocated in the most economically efficient manner, but does not imply equality or fairness. An economy is said to be in a Pareto optimum state when no economic changes can make one individual better off without making at least one other individual worse off. For a review, see Roll (1977); and Schumpeter (1979).

costs, which can undercut bankers' incentives to monitor their loans. This makes investors less willing to lend their money to banks and, when moral hazard is acute, prevents banks from raising funds and lending to non-financial firms. So, despite being the first-best policy to deal with troubled banks, bail-in can have ex ante negative effects. To avoid credit crunch, policy makers should pre-commit themselves to make use of a policy mix, including bail-in together with other resolution mechanisms, such as bail-outs, which can mitigate the adverse effects of bail-in on banks' funding costs and monitoring incentives.

5.2 An 'at risk' debt requirement

In the aftermath of the GFC, many authors have suggested to restructure the liability side of a large bank's balance sheet, so that the un-insured financial liabilities be separated from the equity capital by an amount of long-term debt (Calello and Ervin, 2010; Stein, 2012; and Tarullo, 2013; and McAndrews et al., 2014). They have advocated that large banks should be required to issue a certain amount of long-term, 'at risk' debt, to be converted into equity in resolution. If issued in sufficient quantities, the 'at-risk' debt requirement immunizes the holders of un-insured financial liabilities from losses and can help to reduce their incentives to run.³²

This approach aims to mimic the effects of the normal insolvency procedures, in observance of the no-creditor-worse-off (NCWO) principle, which requires that no creditor shall incur losses in resolution more than it would have incurred in a liquidation scenario. However, the compliance with the NCWO principle is extremely difficult to achieve in a resolution. This is particularly true in situations in which the bail-inable liabilities are governed by laws of different jurisdictions, where insolvency laws are not harmonized.³³

³² The essence of the bail-in is that some creditors should have part of their claims against the bank written down in full or in part, after the write down of lower ranking subordinated claims and equity. Such senior creditors may receive new shares in the bank, but subordinated creditors may have their claims simply extinguished. In terms of order of priority, capital instruments are affected first (common equity tier 1, additional tier 1 and tier 2), followed by: subordinated junior liabilities; uncovered senior liabilities; uncovered deposits; and the DGS the bank is affiliated to for the covered deposits. Each class has to be affected before the following class can be impacted, and within each class, creditors are subject to pro-rata. See Gleeson (2012).

³³ This is the situation prevailing in the EU BU, where the national insolvency procedures are still not harmonized, and where the framework for bank crisis management complies with multiple criteria. Majnoni et al. (2021) suggest that a broader effort is required to streamline the current criteria into a single rulebook, achieving effectiveness mainly through simplification. Its adoption in the BU would help to frame a common approach to failing banks of all sizes

Technically, bail-in is a process which applies to some but not all of the senior creditors of a bank. As outlined above, a bank's bailed-in senior creditors in certain circumstances have part of their claim written down in full or in part, after the write down of lower ranking subordinated claims and equity. Such senior creditors may receive new shares in the bank, but subordinated creditors have their claims extinguished.

An 'at-risk' debt requirement would also have helpful incentive effects, as it would tend to discourage the over-issuance of runnable un-insured financial liabilities. These tools would improve the likelihood that failures could be avoided, or be of a scale that can be managed orderly by authorities. This new requirement serves as capital to be used in resolution, with the objective to limit runs by holders of un-insured financial liabilities. At the end, resolutions turn out to be more frequent, but also to be managed more orderly.³⁴ In this context, the holders of un-insured financial liabilities are less likely to run a bank with both long-term debt and equity in its balance sheet, than a bank having only equity (even in larger amount).

Authorities should consider carefully the reaction function of market participants for these 'at-risk' debt instruments, given the peculiarities of their pricing on the markets. The markets dynamics should be carefully assessed, even before any decision on the issuance of long-term debt is taken. In this respect, the discretionary powers assigned to supervisors in the bail-in process could prevent the holders of bail-in eligible debt to calculate correctly their risk of loss, thus making difficult to price such instruments on the markets. In other words, since the activation of the bail-in trigger and the quantum of the resulting write-down or conversion are in the discretionary domain of the regulators, it would not be possible for the holders of bail-inable debt to make any meaningful estimate of their risk of loss. This would make such debt difficult or impossible to price on the markets.

and would provide a unifying force and a solution to the geographic and institutional fragmentation of the current setup.

³⁴ Some authors explain some reasons in favour of a requirement scaled to the amount of uninsured financial liabilities. This requirement would provide a buffer in resolution to protect the holders of these liabilities; and would avoid a run, thus diminishing the messiness of a large bank's failure. By imposing such a requirement, expected to be costly, the incentives for a reliance on un-insured financial liabilities can be reduced, with the effect to improve the stability of the overall funding of the bank. Finally, tying such a requirement to the amount of the un-insured financial liabilities would force banks to internalize the externalities (the risk of fire sales) stemming from the issuance of short-term, money like liabilities. See: McAndrewes at al. (2014); and Stein (2012).

Since the activation of the bail-in necessarily involves a degree of discretion that cannot be eliminated, the pricing process is facilitated since supervisors have created a credible bank's capital regime, and have given a broad guidance as to in what circumstances they would expect to use their bail-in powers. If a bank is not allowed to operate at the minimum levels of the solvency ratios, the probability of a default decreases, and this situation can be observed, with positive effects for the pricing of the bail-in instruments on the markets.

This objective can be accomplished by either stating contractually that, in the case of need, the creditor's outstanding amount is reduced by his contribution to the required recapitalization, in exchange of the bank's shares, or providing the authorities with statutory powers to achieve the same result (or a 'hybrid' combination of these two methods). However, market participants should take into account that the losses stemming from a liquidation or the dismembering of a bank in a resolution are likely to be higher, and investors in bail-in debt instruments are likely to be better off than under the alternatives of the insolvency or the use of other resolution tools. This could contribute to a more meaningful estimate of losses, in the case of a triggering of a bail-in and the resulting conversion or write-down.³⁵ In sum, it is argued that ensuring creditors losses is an appropriate way to enhance market discipline (Gleeson, 2014).

Box No. 1: Systemic implications of bail-in: a brief survey of the economic literature

The economic literature has extensively studied the long-standing question of how to solve the TBTF problem. The starting point is the GFC, which forced governments to choose between the undesirable alternatives of either bailing out a systemically important bank or allowing it to fail disruptively, with the result that the states intervened in financial markets, and became a major shareholder in many banks.

Morrison (2011) and Bernake (2016) underline that the most important costs of the TBTF problem are incurred in the form of distorted incentives that arise as a consequence of distortions to the capital markets, and to the choice of banks' scale and scope. They argue that it is impossible credibly to withdraw the TBTF policy, and, hence, that it should be managed to minimize the costs of these distortions. A strong role is envisaged for policies aimed at intervening on institutional design, the bank scope, and capital regulations.

³⁵ Under the EU resolution framework, the 'no creditor worse off' safeguard applies equally to the use of the debt write down tool, which means that the likely loss given default on bail-in is at least no worse than the loss given default on other resolution outcomes.

Calomiris (2011) and Philippon and Salord (2017) say that a resolution procedure that requires a minimum haircut on uninsured creditors is essential to promote financial stability. Furthermore, bailins are better than bail-outs, since they soften the negative impact of banks' insolvency on the economy at no cost for tax-payers. Using an agent- based model framework with no moral hazard to evaluate the economic and financial implications of bail-in, Klimek et al. (2015) argue that bailins are always more efficient than bail-outs. They find that for an economy characterized by low unemployment and high productivity the optimal crisis resolution with respect to financial stability and economic productivity is to close the ailing bank. For economies in recession with high unemployment, the bail-in tool provides the most efficient crisis resolution mechanism. Under no circumstances do taxpayer-funded bail-out schemes outperform bail-ins with private sector involvement.

Some studies observe that bail-outs can encourage bankers to take on more risk (Repullo, 2005) and discourage uninsured debt-holders from monitoring the conduct of the bank (Kaufman, 1991), thus leading to moral hazard behaviors. Bail-outs are also seen as a source of systemic risk, in that they give banks an incentive to pick correlated risks as to maximize bail-out gains in the case of a systemic crisis (Farhi and Tirole, 2012). However, other contributions have stressed that bail-outs might avert contagion risks by reducing the likelihood of fire sales and attenuate moral hazard, by increasing the charter value of banks and so inducing bankers to choose a safer portfolio (Acharya and Yorulmazer, 2008, and Cordella and Yeyati, 2003). Freixas (2000) outlines how in order to minimize the losses stemming from the liquidation of a large bank, and at the same time mitigating the moral hazard problem implicit in bail-outs, a mixed strategy would be preferable, in which banks are bail-outed not with certainty but only with a probability less than one (the so-called constructive ambiguity).

Some theoretical approaches have drawn attention to the potential disadvantages of bail-ins. Avgouleas and Goodhart (2015) contend that bail-ins might not be sufficient to reduce the threat of a systemic crisis, when bail-outs might be preferred. They stress that a bail-in mechanism adopted to recapitalize bank as going concern has some advantages and some disadvantages vis-a`-vis a bailout approach. Lower levels of moral hazard, better creditor monitoring, more protection for taxpayers and fairness in placing the burden, lower levels of concentration in the banking sector, improved ex ante behavior of bank management, less sovereign/bank debt 'doom-loop'. Among the latter: more contagious and pro-cyclical; more litigious; slower and more expensive as a process; requiring greater subsequent liquidity injections; leading to deterioration of governance; requiring higher funding costs to banks; providing a worse outlook for bank borrowers and worsening ex post outcomes. Similarly, Dewatripont (2014) discusses the potential pitfalls of bail-ins, concluding that bail-outs should not be excluded ex ante, rather it should complement bail-ins, especially in times of severe crises.

Some economists have modelled contagion effects across market participants (Caccioli et al. (2014); Cont and Schaanning (2017); and Farmer et al. (2020). In these studies, the behavior of banks is

driven by incentives (i.e. increase the return on equity), subject to constraints arising from regulation, contractual obligations, and internal risk limits. A bank is dominated by survival actions to meet these constraints to avoid failure rather than by normal-time activity. Examining the behavior of bank CDS spreads and stock prices, before and after the activation of the bail-in, around the date of the implementation of the BRRD in the EU, Schafer et al. (2016) find empirical evidence of a negative impact of bail-ins on subsequent bank returns, especially for the systemic banks and for the banks in peripheral Eurozone countries. This is due to the increase in funding costs, given the decreased chance of future bail-outs.

Evidence of real effects of the bail-in is provided by Beck et al. (2017), who investigates the reaction of banks exposed as credit suppliers to a bail-ined bank, and find that the banks more exposed cut back more on credit granted to non-financial firms. Taken together these studies show that bail-in increases the cost of debt, due to the shift in the burden of bank failures from tax-payers to debt-holders. Kleinnijenhuis et al. (2021) stress that banks seek to avoid default by aiming to full contractual obligations and comply with regulatory and market-based constraints. They model the interaction between funding contagion, overlapping portfolio contagion and exposure loss contagion in a multi-layered financial system consisting of banks and non-banks. In this context, they find that: financial stability hinges on bail-in design; bail-ins could be a credible tool for resolving failing systemic banks in system-wide crises, only if bail-in is well designed; a crisis-proof bail-in involves an early bail-in, a strong recapitalization and fair conversion rates; an ill-designed bail-in amplifies contagious shocks substantially more than a well-designed one.

Huser et al. (2017) evaluate the systemic consequences of bail-in in the European Union, drawing on a calibrated multi-layered network model of bank debt and equity cross holdings. On this basis, the bail-in of a bank can be simulated to identify the direct contagion risk to the other banks in the network. They find that there is no direct contagion to creditor banks. Spill-overs also tend to be small due to low levels of securities cross-holdings in the interbank network. They also quantify the impact of a bail-in on the different liability holders. In the baseline scenario, shareholders and subordinated creditors are affected by the bail-in, senior unsecured creditors in 75% of the cases.

The conceptual literature that critically evaluates some key aspects of the bail-in design includes the contributions of Zhou et al. (2012), Conlon and Cotter (2014), Sommer (2014), and Persaud (2014). They all stress the importance of an adequate design of the regulatory framework underpinning the application of such a tool. In this context, they stress that the triggers for bail-in powers should be consistent with those used for other resolution tools, and the need to impose minimum requirements on banks for issuing unsecured debt, in order to forestall potential runs by short-term creditors and avert a downward share prices spiral. Similarly, Anderson (2011) also argues that an effective bail-in process requires a careful design of the underlying regulatory framework, especially as regards the allocation of ownership and control rights among the investors involved in the recapitalization.

6. Minimum conditions for an efficient bail-in regime

6.1 The credibility of a bail-in regime

The economic literature points out that a key condition for an efficient application of a resolution tool refers to the careful design of its regulatory framework.³⁶ In the case of the bail-in, a clear and coherent legal framework is essential, and needs to establish an appropriate balance between the rights of private stakeholders and the public policy interest in preserving financial stability. This legal certainty acts as a precondition for the credibility of the bail-in regime, particularly in a cross-border context.

The involvement of the private sector may occur either as part of a resolution that leads to the closure of the bank in distress (gone-concern basis) or under a regime that allows the failing bank to remain open through and after resolution (going concern basis, or open-bank bail-in). ³⁷ Regardless of the legal form in which the bail-in tool can be framed, it is necessary that the losses of the failing bank be allocated precisely to the bank's investors. This because bail-in provisions in essence ensure that a pre-planned contract replaces the bankruptcy process, giving greater certainty as regards the amount of funds able to cover bank losses (Avgouleas and Goodhart, 2015). In this context, debt restructuring ideally would not be subject to creditors' approval, but a NCWO test is introduced to safeguard creditors and shareholders' interests.

The bail-in regime may reach more efficient outcomes than a regime based upon government guarantees and bail-outs, if investors who perceive that bank capital instruments have become more risky are in a condition to decide accordingly. This means that bail-in may not produce the pro-cyclical effects outlined by some economists, ³⁸ if investors can calculate and price correctly

³⁶ For these strands of literature, see: Pennacchi (2010); Calomiris and Herring (2012); Armour (2014); Sommer (2014); Avgouleas and Goodhart (2014); Martynova and Perotti (2015); and Philippon and Salord (2017).

³⁷ The private sector can be involved either by bailing-in parts of the liabilities linked to the sold or transferred operations and sending the original bank into liquidation (the US model), or by fully bailing-in investors in order to achieve a sustainable capital structure for the original legal entity (the EU approach). For more details on the US and the EU bail-in regimes, see infra Box no. 2 and the references there quoted.

³⁸ Lewrick et al. (2019) have studied the pricing of senior bail-in bonds to evaluate the incentives of investors in monitoring banks and pricing bail-in risk. They find that investors price bail-in risk pro-cyclically and this has implications for the design of bail-in regimes. Yet it also implies weaker market discipline on these banks' risk-taking. They say that the flipside of this behaviour is that a tightening in market conditions can trigger a significant rise in the bail-in risk premium, even in the absence of any changes in the issuer's underlying credit risk. Such an increase would push up the banks' cost of funding and, if large enough, could weigh on riskier issuers' ability to roll over their bail-

the bail-in risk, that is the default probabilities, the exposures at default, and the associated loss given default, so that their actual results do not diverge materially from their ex ante expectations. In this case, investors in financial instruments subject to bail-in will not be discouraged and will continue funding other banks. If investors make mistakes, risk premiums are distorted, and the market discipline may send wrong incentives, with abrupt adjustments across markets.

In order to achieve this outcome, it is important that the resolution framework should not add an additional layer of uncertainty, which comes on top of information problems investors naturally face (Gorton, 2009; ICMA, 2017; and Troger, 2018).

Furthermore, bail-in can limit bank runs, and contagion, if it is a credible. For this reason, it is necessary to ensure that the banks have a sufficient amount of bail-inable liabilities, to achieve the internal recapitalization in an orderly, effective and timely manner. This would help reassure the markets that a bail-in would be sufficient to recapitalize the distressed bank, thus forestalling potential runs by short-term creditors and avert a downward price spiral. ⁴⁰

Such instruments could have an impact on banks' financial structure and business model, since they impose the creation and distribution of certain liabilities, thus complicating the balance between rules designed to protect investors and incentives of banks to sell more risky instruments. There could be an increase in the funding costs of banks. However, with a rising proportion of bank creditors at risk from bank failure, there should be a greater benefit in terms of lower funding costs from a safer overall portfolio structure. This because one of the fundamental rationales of an

in eligible debt at a palatable price. This pro-cyclicality highlights the importance of calibrating bail-in regimes conservatively and encouraging banks to issue, in good times, large amounts of bail-in debt across a range of long-term maturities.

³⁹ This is exactly the reason for which the BRRD mandates a minimum requirement of own funds and eligible liabilities (MREL) on the EU banks. The level of the MREL for each bank is set by authorities, depending on size, complexity, interconnectedness, and risk profile of each bank, according to harmonized guidelines issued by the European Banking Authority (EBA). A similar tool, called TLAC (Total Loss Absorbing Capacity) has been established by the FSB since 2019, for the global systemically important financial institutions (G-SIFIs).

⁴⁰ In order to avoid a situation whereby the bail-in is triggered, and the cushion of the bail-inable liabilities is not available, it is needed that they enter into force simultaneously. In this respect, some problems have arisen in the EU: when the BRRD came into force, the bail-in became immediately operative, whereas the completion of the MREL did not.

efficient bail-in regime is that at risk creditors will have ex ante incentives to encourage bank managers to pursue prudent policies, in contrast to more risk-seeking shareholders.

6.2 Conditions for efficiency and policy objectives

The efficiency of the bail-in regime depends on the characteristics of debt holders and their ability to bear losses (Gotz and Troger, 2016). In order to avoid that the bail-in, once implemented, has destabilizing effects on the markets, the capital instruments subject to bail-in are to be held outside the banking sector, by adequately equipped investors, such as insurance companies, pension funds, high-net worth individuals, and hedge funds. ⁴¹ Under these conditions, the failure of a bank may not be the trigger factor for a systemic crisis. If investors who do not possess the required loss-bearing and risk-management capacities are allowed to hold bail-inable capital instruments, in the case of a crisis the likelihood of a bail-out increases, thus undermining the credibility of the bail-in regime itself (Troger, 2018).

The optimal environment for a bail-in to work would be in circumstances where a bank, even a large one, fails for idiosyncratic reasons and the rest of the financial system remains untouched. In the case of exogenous shocks affecting simultaneously a number of banks or financial institutions, the viability of the bail-in mechanism would be determined by the amount of cross-holdings of senior debt across banks. A Zhou et al. (2012) outline that the triggers for bail-in powers should be consistent with those used for other resolution tools. They should be set at the point when a firm has breached the regulatory minima, but before it became balance-sheet insolvent. To make bail-in a transparent tool, its scope should be limited to i) elimination of existing equity shares as a precondition for bail-in; and ii) conversion and haircut to subordinated and unsecured senior debt. This means that debt restructuring under a bail-in should take into account the order of priorities applicable in a liquidation. Furthermore, a well-designated bail-in instrument needs to define a clear-cut, difficult to game trigger event (e.g., a common equity tier

⁴¹ Along the same line of argument, according to Avgouleas and Goodhart (2015), and Hellwig (2021), since the objective of the bail-in is to transfer part of the losses incurred by a bank to its senior creditors, if these senior creditors are other banks, the likely outcome is to transmit contagion across the banking system. In this respect, a properly designed bail-in regime would exclude from its scope deposits, transactions payments swaps, etc., which transfer losses between market participants. Only in the case the bail-in debt is owned predominantly by end investors, the bail-in could substantially reduce systemic risk.

⁴² This is why regulators, especially in the Basel III framework, tend to penalize the inter-bank holdings of debt, and in particular the holdings of other banks' capital instruments.

1 ratio), has to make bail-inable capital instruments identifiable, and has to allow predicting the consequences of the implementation of the tool, i.e., haircuts and conversions should occur automatically with no discretion (Troger, 2018).

Predicting the precise conditions and circumstances for a bank's failure and the actions to be activated by the resolutions authorities is tied to the elements of the so-called Knightian uncertainty, whose distinctive features signal a number of elements and aspects of the real world, include those of the banking crises, which are not susceptible to be measured with precision. ⁴³ A consistent strand of the literature has raised arguments in favor of an implementation of resolution tools with an acceptable level of discretion, within a strategy pursued on a case-by-case basis by the resolution authority (Zhou et al., 2012; Avgouleas and Goodhart, 2014; and Hellwig, 2021). This means that there exists a tension that cannot be entirely eliminated between ex post efficient outcomes and the inefficient ex ante effects of uncertainty, which may undermine the objectives of a bail-in regime, as an adequate mechanism for ensuring that the private sector bear the costs (or part of them) of a banking crisis. ⁴⁴

The discretionary elements in the resolution framework should be limited to the necessary, in order to shield the investors in bail-inable instruments from the negative consequences of the inevitable unpredictability of bank resolutions outcomes. For example, Zhou et al. (2012) recognize the importance to minimize the uncertainty generated by a discretionary use of bail-in powers, in order to avoid surprising and negative effects across markets participants.

In sum, the usefulness of a bail-in regime lies in the fact that by writing off debt it improves the creditworthiness of the bank concerned to a stage where it can access the money markets and raise liquidity. In order to achieve these objective, providers of liquidity must be left without doubt

⁴³ Knight (1921) states that risk and uncertainty are two different categories, because in the former the distribution of the outcomes of a group of instances is known, while this is not true in the case of uncertainty; uncertainty gets in when not all risks are measurable or even knowable. In his view, the complexity of modern finance may imply that risks spread out through the entire system via a number of channels not susceptible of being measured. If we are not in a position to perfectly calculate or assign a probability distribution to the future states of the world, we are not dealing with risk, but with 'Knightian' uncertainty.

⁴⁴ According to Troger (2018), more predictability can be achieved by defining a clear trigger event, independent of the outcomes of the resolution. If the primary goal of bail-in is to induce market discipline exercised by sophisticated investors, the capital layer that absorbs losses does not have to be perfectly adjusted to an individual institution's precise recapitalization needs in resolution. In this way, the scope of bail-in at non-viable banks could be determined ex ante with reasonable certainty, regardless of the strategy authorities decide in an actual resolution event.

that the write-off is immediately effective and cannot be credibly challenged. In a situation where the bank and all its relevant creditors are located in the same jurisdiction, these conditions can be achieved easily. This is not the case for a large bank, whose failure would give rise to significant systemic concerns. In such a situation, a large portion of the bank's senior debt is likely to be governed by laws and contracts other than those of their place of incorporation.

7. Some examples from the EU resolution framework

In the EU the financial architecture for resolution, crisis management and supervision, emerged since the GFC, includes a multi-polar decision making structure, ⁴⁵ and a multi-layered system of applicable rules and standards, at a supranational and national levels. ⁴⁶ In the case of the bail-in, the EU regime appears to be highly prescriptive, but this does not necessarily imply hard rules prescribing with precision the outcomes of the resolution actions. ⁴⁷ One possible explaining factor points towards the degree of complexity of the framework itself, which includes a large number of exceptions, counter-exceptions, and restrictions, which require the use of a significant amount of discretion in the decisions to take by a multitude of responsible authorities. Moreover, the concrete implementation of each rule or standard may depend on the specific circumstances of each case. This means that - once combined with the necessity to coordinate the several layers of authorities and private parties at both the EU and national level - the involved assessments are difficult to make and are subject to varying standards and approaches.

In the analysis of the EU regulatory framework, one has to take into account that in the attempt to reduce as far as possible the space for political decisions, it has been entrusted all to technical bodies, which have to move within defined parameters. The point is the excessive rigidity of the whole process, which cannot be overcome without recourse to discretion and 'constructive

⁴⁵ For more details on the EU financial architecture, see: Pisani-Ferri et al. (2012); Goyal et al. (2013); Pagano (2016); and Philippon and Salord (2017).

⁴⁶ In the EU, the BRRD and the SRMR provide the fundamentals for a harmonized system for crisis management and resolution, at least for the largest and systemic EU banks. The EU framework still misses an efficient and integrated system for the management of crises of small and medium-sized banks. On this point, see Majnoni et al. (2021).

⁴⁷ It is not uncommon that a detailed legal framework is accompanied by elements of discretional choices, and even standardized operational procedures leave substantial room to the relevant official decision-makers for ex post variation of the key decisions, and this tends to put in question its automaticity and uniformity of application (Hadjiemmuil, 2015).

ambiguity'. This is difficult to rely on entities other than those with political steering powers, which are responsible for the decision on the use of public resources and can act through the law.

The sub-paragraphs below draw some examples from the EU resolution framework where the minimum conditions for an efficient bail-in regime, as outlined previously, might not be achieved. They belong to the following areas: prediction of resolution outcomes by investors; preconditions for resolution; selection from the resolution tools; exemption of liabilities from bail-in; restrictions on holdings of bail-in capital.

7.1 Prediction of resolution outcomes

The EU and national authorities exercise their powers to pursue the goals established by the BRRD, in order to facilitate a solution considered preferable in the light of a notion of 'public interest', when compared with the outcomes of a normal insolvency procedure. ⁴⁸

In the application of the bail-in, investors have to base their risk assessments on the administrative practices followed by the different competent authorities. In order to ensure an adequate accuracy of any calculation on the implied default probabilities and loss given default, they have to know who makes the final decision in each step of the procedure. This situation might add significant uncertainty in assessing the timing and the impact of a bank's resolution. The predictability of the resolution outcomes might be impeded in case the distribution of competences seems to be not entirely consistent.⁴⁹

Another source of uncertainty refers to the issue of the pricing of the newly issued bail-inable capital instruments. The high degree of discretion embedded in the regulation renders the level of the risk associated with these instruments (and the related price) almost unpredictable. Any adjustment of the MREL framework affects the pricing of already issued securities and adds another layer of complexity to the existing rules.

⁴⁸ These goals include: i) ensuring the continuity of critical functions; ii) avoiding significant adverse effects on financial stability; iii) avoiding or minimizing recourse to public funds and tax-payers' money; iv) and protecting deposits, investments, and other clients' funds. See BRRD, article 31. See: Cassese (2017); Ventoruzzo and Sandrelli (2020).

⁴⁹ For instance, the EU Commission, primarily responsible for competition, has the power to assess both the compatibility of public intervention with the State aid rules, and the respect of the BRRD requirements referring to financial stability. Additional tension can also emerge between national governments, which are inclined to provide public support to local banks and the compliance with the State aid rules determined at the EU level (Troger, 2018; and Ventoruzzo and Sandrelli, 2020).

7.2 Triggers for resolution

The condition for the activation of the bail-in is that a formal resolution proceeding is triggered by the resolution authority, who decides based upon three criteria. Each of these criteria requires complex supervisory or resolution actions, whose results cannot be calculated with precision ex ante. In this context, the prediction of the outcomes by investors and markets could be hampered. Enhanced certainty can be achieved over time, if there is a reasonable number of cases where the application of the criteria has been done consistently. This can turn out to be particularly difficult in the EU, given the overlap of competences among authorities and the lack of transparency for no-action decisions. Therefore, there could be the risk that this regulatory framework, even with more stringent regulations, could not produce a reduction of the forbearance. St

Under the first condition for resolution, the EU framework specifies the circumstances under which the assessment for a bank to be failing or likely to fail is justified. ⁵² Besides the classical triggers for insolvency, the EU framework states that a bank's failure can occur if the violations of prudential banking regulations, particularly those concerning the own funds requirements, justify the withdrawal of the authorization, or an extraordinary public support is needed. In these cases, uncertainty remains important, since these triggers deviate from the established understanding of a bank's default, which can shape market expectations, given that they cannot be drawn from the practices developed in the national insolvency procedures. ⁵³

⁵⁰ In the EU framework, these three criteria are the following: i) the bank is failing or is likely to fail; ii) there is no (private or public) alternative to the bank's failure; iii) resolution pursues the public interest better than an ordinary insolvency procedure. Article 32 of the BRRD provides the detailed rules governing these conditions.

⁵¹ Avgouleas and Goodhart (2015) doubt that the adoption of bail-in regimes would lead to earlier regulatory interventions than under the bail-out regimes. They point out that authorities have the incentives to delay intervention because of higher legal risks in the case losses are imposed on bank creditors, which is not present in bail-out scenarios. These legal concerns could unduly delay resolution until the last possible minute. By then the liabilities needed to be written down could extend beyond the contract bail-inable debt and affects a wider range of creditors. There will be more parts to the negotiation and hence that may be more protracted proceedings.

⁵² For an overview of the issues involved in such an assessment, see the following: Zhou et al. (2012); Wojcik (2016); and Busch et al. (2019).

⁵³ According to the BRRD, the classical insolvency triggers refer to a situation where there are objective elements to assess that in the near future: i) the assets of the bank will be less than its liabilities, or ii) the bank can be unable to pay its debts or liabilities as they fall due. Even for these classical triggers, there is an issue of uncertainty. These elements cannot be unequivocally determined especially under serious time constraints. The assessment of non-

These difficulties gain prominence, if one takes into account the exceptions defined in the EU law to the general principle that government support automatically triggers resolution. On the one hand, the EU framework requires that the public support (in various forms⁵⁴) to an ailing bank be extended to avert a systemic crisis. On the other, these measures have to be precautionary and temporary, finalized to assist solvent banks, and are not supposed to offset losses the bank has incurred or is likely to incur in the near future. Things become even more complicated, if one takes into consideration that these precautionary recapitalizations are allowed to cover capital shortfalls revealed in hypothetical adverse scenarios used in the EU or national stress test exercises. This means that the exception as defined in the EU law requires to assess the systemic implications of a hypothetical failure of a bank that is still going-concern. On the contrary, only private funds can cover shortfalls emerging under the baseline scenario.⁵⁵

The identification of the banks for which government support without minimum bail-in is available is not an easy task for investors seeking to assess the risk inherent in an investment in bank's bail-inable capital instruments. For this purpose, investors have to understand when a failure of a bank has the potential to determine a systemic crisis. This cannot be derived by the SSM designation of a bank as significant. A more granular analysis is needed to look at the position of each bank within a Member state, or the European or the global economy. The issue here is that market participants usually are not in a position to assess with precision all these elements, since they do not have all the necessary information.

viability is extremely difficult, and is very discretionary, given that these resolutions conditions require the knowledge of elements not publicly available, such as the key data and the relevant methodologies used by authorities and the specification of the concept of the 'near future' referred to in the law and used in the actual cases. For these considerations, see Troger, 2018.

⁵⁴ The BRRD refers to the following forms: a state guarantee to: i) liquidity facilities by central banks; or II) newly issued liabilities; iii) an injection of own funds or purchase of capital instruments, with no advantage for the bank.

⁵⁵ Moreover, public support must comply with State aid rules at both the EU and national levels. In this respect, a very important condition concerns the so-called burden sharing, i.e., the need to access public sources that existing eligible shareholders or creditors contribute to the financial effort to restore the bank's health. More specifically, resolution funds can be accessed only after an amount equal to 8 per cent of liabilities has been covered through write-downs or conversions of liabilities and, in any case, remaining losses can be covered only to a maximum of 5 per cent of the bank's liabilities. See the articles 37, 45, 56, 57, 58 of the BRRD.

7.3 Selection of the resolution tools

In the EU framework, once resolution has been triggered, the activation of the bail-in is not automatic. Resolution authorities may choose the resolution tools that are more coherent with the resolution strategy adopted for each individual bank from a set of tools defined by law.⁵⁶ This means that even in a resolution the involvement of the private sector is not mandatory, given that the resolution authorities may opt to manage the crisis applying the other tools.⁵⁷ In this respect, resolution authorities have a significant level of flexibility as to the choice of the tool to use in a crisis scenario.

Investors who seek to anticipate the outcomes of a resolution face a wide array of possible choices not only on the levels of the selection of the appropriate strategy and tools, but also on the level of their implementation. This prediction is even more complicated given that the EU regime only partly supra-nationalises the relevant decisions, and the division of responsibilities between and within each layer foresees a number of overlapping competences and interests in designing and implementing rules and procedures. Accordingly, the outcomes of a resolution may vary to a certain degree as a function of the level of the decision-maker. In this context, the lack of the minimum (theoretical) conditions for an efficient bail-in regime (legal certainty, and prediction of outcomes by investors) matters, given that the degrees of discretion available to authorities may lead to diverging outcomes, thus increasing uncertainties for market participants.

7.4 Liabilities exempt from bail-in

In the EU resolution framework, there are liabilities that are either exempt from bail-in directly by law, or are excluded in exceptional circumstances at the discretion of the resolution

⁵⁶ The resolution tools foreseen in the BRRD (article 37) are the following: the sale of business tool; the bridge institution tool; the asset separation tool; the bail-in tool. Resolution authorities may apply the resolution tools individually or in any combination. In the US, the Federal Deposit Insurance Corporation (FDIC), which leads the special resolution regime created by the DFA has also broad discretionary powers in selecting the appropriate resolution tools. For an overview of the advantages and disadvantages of each specific tool, in the context of a compared analysis of the US toolbox, see: Zhou et al. (2012); Busch et al. (2019); and Ventoruzzo and Sandrelli (2020).

⁵⁷ The only constraint established in the BRRD (article 37.2) is that if the chosen resolution strategy leads to the involvement of creditors, AT1- and T2-instruments have to be written down or converted immediately, regardless of when the bail-in tool is applied.

authorities. In the first case,⁵⁸ their identification seems to be not straightforward, because some of the terms used require an interpretation and an in-depth knowledge of the bank's business model, organizational structure and investments decisions. The same holds for the liabilities excluded by the authorities.⁵⁹ It could be difficult to forecast, at the time of the investment, whether certain liabilities are distributed in a way that their bail-in would lead to run-like scenarios or impede the continuity of critical functions of the bank.⁶⁰

These activities require a knowledge of the methodologies, and data, used by regulators and banks, which are not available to external actors. Moreover, many investors are not sophisticated enough to exercise such an informed monitoring function. These difficult predictions are relevant not only for those creditors who potentially benefit from the exceptions. Also other creditors, who are equally subject to bail-in, are in a position to bear additional costs, since granting an exception implies that the loss participation of other creditors increases. ⁶¹

The degree of uncertainty stemming from the current regime of statutory or discretionary exceptions can be reduced. An approach aimed at limiting the bail-inable liabilities to a sufficiently thick layer of predetermined liabilities would reduce the uncertainty, facilitating the investors' predictions about the magnitude of the private sector involvement in a crisis. Under such a regime, only investors in these capital instruments would exert market discipline, because pending losses

⁵⁸ The liabilities excluded by law refer to: covered deposits; secured liabilities, including covered bonds and liabilities for hedging purposes; clients' assets and funds, held as a separate asset protected by the law; liabilities from a fiduciary relationship; liabilities towards third party financial institutions and payment or settlement systems, with a maturity of less than seven days; liabilities to commercial creditors, providing critical goods or services; liabilities towards employees for remuneration and benefits; liabilities to tax and social security authorities, preferred by law; and liabilities towards deposit guarantee systems, arising from contributions. See article 44.2 of the BRRD.

⁵⁹ The liabilities excluded at the discretion of the resolution authorities (article 44.3 of the BRRD) refer to the following situations and motives: practical impossibility of timely bail-in; continuity of critical functions; prevention of bank runs; avoiding destruction of value. Discretionary exemptions must comply with the no-creditor-worse-off (NCWO) principle, since any exemption of certain liabilities from possible haircuts might worse the position of other bail-inable creditors.

⁶⁰ Based upon the evidence suggested by the GFC, Gorton (2009) has explained that parallel bank-like runs can occur outside the traditional banking system in wholesale markets, triggered by sudden information regarding the misperceived risk-structure of certain asset classes.

⁶¹ For a discussion of the rationale in favour of excluding certain types of liabilities from bail-in, and the related financial stability implications, see: Zhou et al. (2012); Avgouelas and Goodhart, 2014; Wojcik (2016); Philippon and Salord (2017); and Troger (2018).

were far more assessable. The gains in market discipline should over compensate the losses in additional monitoring and militate in favour of limiting the bail-in to a designated part of the liabilities side of the bank balance sheet (Gotz et al., 2017).

7.5 Restrictions on holdings of bail-in capital

The effect of the bail-in mechanism is to allocate some of the losses incurred by a bank to its uninsured creditors and shareholders. If they are other banks or financial institutions, then the likely effect of a bail-in implementation could be the transmission of contagion across the financial system. In other words, the application of bail-in should not spread instability from one bank to another, since this may trigger a systemic crisis. This is why the post-GFC prudential framework includes provisions aimed at deducting such investments from the banks' regulatory capital.

We have seen before that a properly designed bail-in regime minimizes this risk, by excluding from its scope the transactions types, which transmit losses between the financial institutions (deposits, transactions payments, swaps, etc.). Enhanced market discipline is pursued by imposing constraints on the demand side for bail-able capital instruments. Such investors should have the necessary loss bearing capacity, when their debt is written down; have the capacity to hold bank equity in the case of a conversion; understand and price correctly the risk of bail-in; and charge adequate risk premiums on the new capital instruments, thus exerting a significant monitoring activity on banks' behaviour. If these instruments are allocated to unsophisticated (retail) investors, market discipline could be reduced, with some consequences on the credibility of a bail-in regime. These considerations have offered arguments in favour of possible restrictions in the holdings of bail-inable instruments.

The economic literature has outlined some qualifications of the suitable investors, who should be among sophisticated financial institutions, outside the banking sector, and with no asset-liability mismatch.⁶² If bank capital is held by 'wrong investors' the private sector involvement may produce unintended consequences, which in turn may induce decision makers to use the

⁶² Gotz and Troger (2016) have highlighted that a bail-in of debt holders becomes especially problematic if the debt holders are households or other types of retail investors. So, they argue that bail-in is most effective if such debts holders are (i) sophisticated investors which are (ii) active outside the banking sector and (iii) are not subject to an asset-liability mismatch due to their investment strategy. To this end, in their view, supervisors should continuously gather information on the owners of banks' bail-inable debt and monitor the ownership structure to assess whether debt holders satisfy the above-mentioned three criteria.

discretion built in the regulatory framework to restrict the application of the bail-in tool, making the involvement of the private sector essentially a politically driven event.

In general, the question of the desirable nature of holders of bail-inable instruments is difficult to solve. As noted above, retail investors are not suitable for reasons of poor monitoring aptitude and risk of panic. Other banks are to be excluded for reasons of contagion. The point is that other financial institutions, other than banks, can also give rise to episodes of systemic contagion: the notion cannot be restricted to the banking system, also because banks, insurance companies and asset managers are often closely interconnected, Super-rich individuals and sovereign wealth funds remain. These considerations are not an argument against the imposition of bail-inable securities, but against an excessive faith in the ability of this instrument to resolve the classic trade-offs related to bank defaults (moral hazard and the burden on public finances on the one hand, systemic risk on the other).

Box No. 2: Some aspects of the US and EU bail-in regimes 63

In the US, until 2010, insolvent depository banks, were (and are) subject to a special federal administrative procedure regulated by the Federal Deposit Insurance Act (FDIA). Non-bank financial companies, including bank holding companies, were subject to ordinary (court-based) insolvency proceedings under the US Bankruptcy Code. In 2010, the Dodd-Frank ACT (DFA) introduced a new administrative option (the 'orderly liquidation authority', OLA) for non-bank financial companies recognized as systemically important financial institutions. As with depository banks, the DFA grants FDIC special powers to assist a failing institution.

The US resolution authority is assigned a significant amount of discretion in the application of the general principles concerning the equal treatment of creditors and the pecking order, in the event of both a bail-in and other resolution tools. In both contexts, resolution is a procedure managed by an administrative authority (the SRB and the national resolution authorities in the EU; the FDIC in the US), as opposed to a bankruptcy court.

As to the scope of application, the EU includes all banks and investment firms, as well as financial holdings companies, while the OLA carves the insured depository institutions out. As to resolution authorities, the US shows a streamlined structure, where the FDIC operates as single federal resolution authority. The complex EU framework reflects a fragmented institutional context, where significant banks are supervised at the ECB level, while the others at the national level. Another

⁶³ This Box is derived mainly form the following papers: Sommer (2014); Ringe (2016); Avgouelas and Goodhart (2015); and Ventoruzzo and Sandrelli (2020).

difference concerns the resolvability at a group level and the related adoption of the single point of entry (SPE) vs the multiple point of entry (MPE) approach. The MPE strategy is the rule in the EU, where a holding company may be resolved even if it does not meet the resolution requirements, if one or more subsidiaries do meet them. The US adopts largely a SPE model, where the resolution triggers are checked at the holding company or at the subsidiary level.

The bail-in is different in scope in the two systems. In the EU, an 'open bank' bail-in is theoretically practicable. Resolution authorities may apply the bail-in tool to recapitalize a failing bank to restore its long-term viability. In the US, the bail-in tool may only be used in a gone-concern situation, possible coupled with the sale of business or assets to a bridge institution or a third party, where the resolved bank ceases to exist ('closed bank' bail-in).

Finally, in the US, the resolution mechanism interplays with the federal insolvency rules, both ordinary (the Bankruptcy Code) and special (the FDIA). In the EU, since the winding-up and liquidation of banks are not harmonized, a European resolution decision reflects the intersection with different national rules, and this situation may determine a number of potentially divergent outcomes.

8. Does bail-in end the too-big-to-fail problem?

Since its inception, the bail-in has been at the center of academic and professional disputes as regards its effectiveness to mitigate the systemic risks associated with disorderly liquidations. While bail-in has proved to be relatively successful in dealing with the failure of small and medium-sized banks, its ability to handle system-wide crises has not been tested. In this respect, its ability to end the TBTF problem needs to be confirmed. Accordingly, the issue whether it may replace the government bail-outs, dropping out the implicit governments' guarantees in the price of large banks' capital, should be assessed carefully.

In general, the post-GFC bail-in approach seems to be based on the assumption that banks fail individually, one at a time. On the contrary, the historical experience tells us that bank failures are rarely pure idiosyncratic events, since they tend to happen simultaneously, essentially when there is a wider economic distress. In the case of a system-wide crisis, it could be unrealistic to believe that bail-in can be activated on a magnitude that allows absorbing the losses, or ensuring the recapitalization needs of all the involved banks simultaneously. In particular, there is the risk that imposing haircuts on a large bank's creditors cannot preserve financial stability, but rather it can accelerate a systemic panic. The situation is even more complicated since, in this scenario, alternative private sources of financing tend to be scarce. Similar arguments hold also in the case

of an individual failure affecting any of the largest and systemic banks. ⁶⁴ According to some authors, bail-in could be used to recapitalize even a large bank, but only if it has failed due to its own actions and not due to a generalized systemic crisis, i.e., its failure is clearly idiosyncratic; otherwise, a flight of creditors from one bank to others may be uncontainable (Avgouleas and Goodhart, 2015). ⁶⁵

These arguments have been at the centre of the decisions taken in the post-GFC years by the competent authorities when they have been faced with the issue to apply (or more often not to apply) the bail-in tool (for an extensive review of the several cases incurred in the EU see Ventoruzzo and Sandrelli, 2020). The last episode in order of time has been the crisis of the Credit Swiss in 2023: it shows all the difficulties and problems that have to be dealt with in the case of a possible application of the bail-in to a large and systemic bank.

The problems of Credit Swiss were found in a range of business areas and in various risk types. In almost all cases, serious deficiencies in risk management played a role. FINMA's measures targeted these deficiencies and tightened up checks and controls. Despite some extensive adjustments over the years, the bank's executive bodies were unable to remedy the repeatedly identified deficiencies in the bank's organisation in a holistic and sustainable manner. The bank met the regulatory capital requirements. Even this capital adequacy situation was unable to contain or prevent the massive crisis of confidence. It also met the regulatory requirements for liquidity and held comfortable liquidity buffers in summer 2022. However, the loss of confidence in the bank led to very rapid and widespread liquidity outflows, which were exacerbated by digital means of communication (digital bank run) and ultimately brought the bank to the brink of insolvency.

During the resolution weekend, the Swiss authorities had considered 4 options to solve the crisis of the Credit Suisse. The first one was a bail-in resolution of the bank. The Swiss authority would have declared the point of non-viability and ordered restructuring and capital measures. This was a TBTF resolution option and had been prepared by the international crisis management

⁶⁴ For example, it could be questioned whether in the Banking Union the total amount of resources available on ex ante basis to all the deposit guarantee systems (DGS) and the single resolution fund (SRF) may be enough in the case of a crisis affecting one of the BU largest banks. These funds are clearly insufficient in the case of a system-wide banking crisis. For these calculations and assessment, see Hadjiemmanuil (2015).

⁶⁵ This holds above all if the concerned bank is a pure domestic bank, since in a cross-border context, the multiple layers of risks and costs associated with a cross-border resolution make the use of the bail-in highly uncertain in its outcomes.

group. But it would have required an emergency law to secure funding in resolution. The second option - a nationalization and temporary public sector ownership – was dismissed since it is not foreseen in the Swiss TBTF regime. The third option was an assisted merger of Credit Suisse with UBS. The fourth was a bankruptcy and activation of the Swiss emergency plan. In the end the option chosen was the merger which was considered as the least risky option. ⁶⁶

For the above considerations, the bail-in tool seems to be more viable in the case of a crisis affecting a bank whose dimension and risk profile could not give rise to systemic concerns. A successful bail-in recapitalization would require a quick restoration of market confidence, accurate evaluation of losses and successful restructuring of the bailed-in bank's operations. It could be very hard for regulators to secure all these prerequisites in the event of a systemic crisis. A complete replacement of the bail-out subsidy is very difficult to obtain, because the likelihood of a public bail-out and the related estimated value of the government guarantee are not the same for all banks nor invariant over time. In particular, they tend to increase with banks' size and with the deteriorating macroeconomic conditions.

We have learnt from history that in the case of a systemic crisis the decisions taken by governments to avoid any backstop to the financial system usually tends to aggravate the consequences of the crisis.⁶⁷ The resolution of a large and systemic bank would still require a credible fiscal backstop.⁶⁸ In this respect, it could be recalled that it is not always true that the interests of tax-payers are in all cases impaired by the governments interventions. Their indirect costs, in terms of reduced market discipline or enhanced moral hazard, have to be assessed against their indirect benefits, such as the economic and financial stabilizing effects from a large bank's rescue. Both costs and benefits cannot be measured with accuracy ex ante. The taxpayers'

^{66.} For more details on the crisis of Credit Suisse, see: Angeloni et al. (2024); and FINMA (2023).

⁶⁷ The tough policy stance of the Federal Reserve System in the run-up and during the Great Depression during the 1930s to allow monetary contractions also by allowing banks to fail has been one of the main contributing factor of the excessive severity and duration of the crisis. See, for example, Eichengreen (1992), and Romer (1993). The same can be stated for the systemic consequences of the GFC, given that the US authorities in 2008 decided to not activate a credible public backstop in the case of the failure of Lehman Brothers. On this last point, see De Bonis and Trapanese (2023), and the references there quoted.

⁶⁸ The EU framework for crisis management still misses a financial stability facility/exemption, aimed at overcoming the rigidities of the framework in the case exceptional circumstances threaten the EU financial stability. This feature, coupled with the absence of a true European deposit insurance system, makes the EU crisis management framework still incomplete. See Trapanese et al. (2024).

estimated losses are higher at the point of the crisis, but it is not uncommon that the investments of tax-payers in bank capital may be reversed overtime. The potential losses for the public participation in bank recapitalization may turn into substantial gains, as the economic conditions improve and the value of the bailed-out bank increases on the markets. ⁶⁹

9. Conclusions

The objectives of the post-GFC bail-in approach (enhancing market discipline, reducing moral hazard, and protecting public finance) can be achieved to the extent market participants can make informed predictions about the authorities' reaction function in a crisis, and foresee the risk of their capital instruments being written-off or converted with reasonable certainty.

The paper shows that in the case of the EU the bail-in is based upon a regulatory framework where the automaticity of the rules is complemented by a significant discretion. If these degrees of discretion are framed in a decision-making process based upon complex mechanisms, there could be uncertainties in the application of the bail-in and difficulties in the achievement of its original objectives. The build-up of high-quality financial instruments, ideally mainly in terms of subordinated debt (to avoid possible overlaps with capital requirements) to be bailed-in may contribute to enhance the predictability of who is supposed to bear the costs of an intervention, and help to solve the associated uncertainties.

From a functional perspective, the paper shows that the bail-in is supposed to work efficiently, if its implementation does not create the conditions for mounting threats to the stability of the financial system as a whole. In conclusion, the paper argues that bail-in should be considered as one of the elements of a comprehensive solution to the TBTF problem.

⁶⁹ For these considerations, see Laeven (2011); Hertig (2012); and Dewatripont (2014).

Bibliography

- Acharya V., and Yorulmazer T., (2008), "Cash-in-the-Market Pricing and Optimal Resolution of Bank Failures", Review of Financial Studies, Vol. 21.
- Acharya V., Shin H. S., Yorulmazer T., (2011), "Crisis Resolution and Bank Liquidity", The Review of Financial Studiers, Vol. 24.
- Adrian T., Ashcraft A. B., (2012), "Shadow Banking Regulation", Federal Reserve of New York Staff Reports, No 559.
- Afonso G., Santos J. A. C., and Traina J., (2014), "Do 'Too-Big-To-Fail' Banks Take on More Risk?", FRBNY, Economic Policy Review, Special Issue, March.
- Anderson R. W., (2011), "Ownership and Control in Bail-in and Special Resolution",
 Discussion note, London School of Economics, FMG Workshop on the Design of Bail-in
 Arrangements and Special Resolution Regimes.
- Angeloni I., Claessens S., Seru A., Steffen S., Weder di Mauro B., (2024), "Much Money, Little Capital, and Few Reforms. The 2023 Banking Turmoil", Geneva Reports, No. 27.
- Armour J., (2014), "Making Bank Resolution Credible", ECGI, Law Working Paper N. 244.
- Avgouleas E., and Goodhart C., (2014), "A Critical Evaluation of Bail-ins as Bank Recapitalisation Mechanisms", CEPR Discussion Paper No. 10065.
- Avgouleas E., and Goodhart C., (2015), "Critical Reflections on Bank Bail-ins", Journal of Financial Regulation, Vol. 1, Issue 1, March.
- Aschcraft A., (2005), "Are Banks Really Special? New Evidence from the FDIC-Induced Failure of Healthy Banks", American Economic Review, Vol. 95, No, 5, December.
- Baltensperger E., (1980), "Alternative Approaches to the Theory of the Banking Firm", Journal of Monetary Economics, Issue No. 6.
- Beck T., Da-Rocha-Lopes S., and Silva A., (2017), "Sharing the Pain? Credit Supply and Real Effects of Bank Bail-ins", CEPR Discussion Paper No. 12058.
- Bernanke B. S., (1983), "Non-monetary Effects of the Financial Crisis in the Propagation of the Great Depression", American Economic Review, Vol. 73, No. 3.
- Bernanke B. S., and Blinder A. S., (1988), "Is It Money or Credit, or Both, or Neither?, Credit, Money, and Aggregate Demand", American Economic Review, Vol. 78, Issue No. 2.
- Bernanke B. S., (2016), "Ending 'too big to fail': What's the right approach?", Brookings Institution's Blog.
- Brandao-Marques L., Correa R., and Sapriza H., (2013), "International Evidence on Government Support and Risk Taking in the Banking Sector", IMF Working Paper No. 94

- Brunnermayer M., (2009), "Deciphering the Liquidity and Credit Crunch 2007-08", NBER Working Paper Series, No. 14612.
- Busch D., van Rijn M. B. J., and Louisse M., (2019), "How Single is the Single Resolution Mechanism?", European Banking Institute, Working Paper Series, Issue No. 30.
- Caccioli F., Shrestha M., Moore C., and Farmer J. D., (2014), "Stability analysis of financial contagion due to overlapping portfolios", Journal of Banking & Finance, Vol. 46.
- Calello P., and Ervin, W., (2010), "From Bail-out to Bail-in", The Economist, January, 28.
- Calomiris C. W., (2011), "Incentive-Robust Financial Reform", Cato Journal, Vol. 31.
- Calomiris C. W., and Herring R. J., (2012), "How to Design a Contingent Convertible Debt Requirement That Helps Solve Our Too-Big-to-Fail Problem", Journal of Applied Corporate Finance, Vol. 25, Issue 39.
- Cassese S., (2017), "A new framework of administrative arrangements for the protection of individual rights", ECB Legal Conference 2017: Shaping a New Legal Order for Europe: A Tale of Crises and Opportunities", Frankfurt, September.
- Ciocca P., e Sannucci V., (1990), "Henry Thornton, primo teorico della banca centrale", prefazione a Henry Thornton, "Indagine sulla natura e sugli effetti del credito cartolare in Gran Bretagna, Torino, Banca CRT.
- Coffee J. C., Jr, (2010), "Bail-Ins Versus Bail-Outs: Using Contingent Capital to Mitigate Systemic Risk", Columbia Law and Economics, Working Paper No 380.
- Conlon T., and Cotter J., (2014), "Anatomy of a bail-in", Journal of Financial Stability, Vol.15.
- Cont R., and Schaanning E. F., (2017), "Fire sales, indirect contagion and systemic stress testing", Norges Bank Working Paper, No. 2.
- Cordella T., and Yeyati E. L., (2003), "Bank Bail-outs: Moral Hazard vs. Value Effect", Journal of Financial Intermediation, Vol. 12.
- De Bonis R., and Trapanese M., (2023), "Le quattro età della regolamentazione: che fare oggi?", Banca d'Italia, Questioni di Economia e Finanza, No. 796, settembre.
- Dewatripont M., (2014), "European Banking: Bailout, Bail-in, and State Aid Control", International Journal of Industrial Organization, Vol. 34.
- Diamond D. W., (1984), "Financial Intermediation and Delegated Monitoring", Journal of Finance, Vol. 38.
- Diamond D. W., and Dybvig P. H., (1983), "Bank Runs, Deposit Insurance, and Liquidity", Journal of Political Economy, Vol. 91, Issue 401.

- Eichengreen B., and Ruehl C., (2000), "The Bail-in Problem: Systematic Goals, Ad Hoc Means", NBER, Working Paper No. 7653, April.
- Eichengreen B., (1992), "The Origins and Nature of the Great Slump Revisited", The Economic History Review, Vol. 45.
- Fama E., (1985), "What's Different About Banks?", Journal of Monetary Economics, Issue No. 15.
- Farhi E., and Tirole J., (2012), "Collective Moral Hazard, Maturity Mismatch, and Systemic Bail-outs", American Economic Review, Vol. 102.
- Farmer J. D., Kleinnijenhuis A. M., Nahai-Williamson P., and Wetzer T., (2020), "Foundations of system-wide stress testing with heterogeneous institutions", Bank of England Staff Working Paper No. 861.
- Financial Stability Board, (FSB), (2010a), "Reducing the moral hazard posed by systemically important financial institutions", October.
- FSB, (2010b), "Intensity and Effectiveness of SIFI Supervision. Recommendations for enhanced supervision", November.
- FSB, (2011), "Policy Measures to Address Systemically Important Financial Institutions", November.
- FSB (2013), "Recovery and Resolution Planning for Systemically Important Financial Institutions. Guidance on Developing Effective Resolution Strategies", July.
- FSB, (2014), "Key Attributes of Effective Resolution Regimes for Financial Institutions", October.
- FSB, (2015), "Principles on Loss-absorbing and Recapitalization Capacity of G-SIBs in Resolution, Total Loss-absorbing Capacity (TLAC) Term Sheet", November.
- FSB, (2019), "Review of the Technical Implementation of the Total Loss-Absorbing Capacity (TLAC) Standard", July.
- FINMA, (2023), "The lessons learned from the crisis of the Credit Suisse", December.
- Freixas X., (2000), "Optimal Bail Out Policy, Conditionality and Constructive Ambiguity", DNB Staff Reports, Netherlands Central Bank.
- Gadanetz B., Tsatsaronis K., and Altunbas Y., (2012), "Spoilt and Lazy: The Impact of State Support on Bank Behavior in the International Loan Market", International Journal of Central Bank, Vol. 121, Issue 8.
- Gleeson S., (2012), "Legal Aspects of Bank Bail-ins", LSE Financial Markets Group Paper Series, Special Paper No. 205.

- Goodhart C. A. E., (1987), "Why Do Banks Need a Central Bank?", Oxford Economic Papers, Vol. 39.
- Goodhart C.A.E., (1989), "Money, Information and Uncertainty", London, Macmillan.
- Gordon J. N., and Ringe W. G., (2015), "Bank Resolution in the European Banking Union: A Transatlantic Perspective on What It Would Take", Colombia Law Review, Vol. 115.
- Gorton G., and Metrick A., (2012), "Securitized Banking and the Run on Repo", Journal of Financial Economics, Vol. 104.
- Gorton G., (2009), "Information, Liquidity, and the (Ongoing) Panic of 2007", American Economic Review, Vol. 99.
- Gotz M. R., and Troger T. H., (2016), "Should the Marketing of Subordinated Debt Be Restricted/Different in One Way or the Other? What to Do in the Case of Mis-Selling?", In-Depth Analysis for the Economics and Monetary Affairs Committee of the European Parliament, No. 6.
- Gotz M. R., Krahnen J. P., and Troger T. H., (2017), "Five Years after the Liikanen Report: What Have We Learned?", SAFE Policy Paper No. 50.
- Goyal R., Brooks P. K., Pradhan M., Tressel T., Dell'Ariccia G., Leckow R., and Pazarbasioglu C., (2013), "A Banking Union for the Euro Area", IMF Staff Discussion Note, No 01.
- Gropp R., Hakenes H., and Schnabel I., (2011), "Competition, Risk-Shifting, and Public Bail-Out Policies", The Review of Financial Studies, Vol. 24.
- G-20 (2009), "The Pittsburgh Summit G-20 Leaders Declaration".
- Hadjiemmanuilm C., (2015), "Bank Resolution Financing in the Banking Union", London School of Economics, LSE Law, Society and Economy Working Paper No. 6.
- Hellwig M., (2021), "Twelve Years after the financial Crisis Too-big-to-fail is still with us", Journal of Financial Regulation, January.
- Hertig G., (2012), "Governments as Investors of Last Resort: Comparative Credit Crisis Case-Studies", European Corporate Governance Institute, Law Working Paper, No. 187, January.
- Huser A.C., Halaj G., Kok C., Perales C., and van der Kraaij A., (2017), "The systemic implications of bail-in: a multi-layered network approach", Journal of Financial Stability, Vol. 38.
- Kalemli-Ozcan S., Sorensen B., and Yesiltas S., (2012), "Leverage Across Firms, Banks, and Countries", Journal of International Economics, Vol. 88, Issue 2, November.
- Kaufman G. G., (1991), "Lender of Last Resort: A Contemporary Perspective", Journal of Financial Services Research, Vol. 5.

- Kleinnijenhuis A., Goodhart C. A., and Farmer D., (2021), "Systemic Implications of the Bailin Design", CEPR, Discussion Paper No. 16509.
- Klimek P., Poledna S., Farmer J. D., and Thurner S., (2015), "To bail-out or to bail-in? Answers from an agent-based model", Journal of Economic Dynamics and Control, Vol. 50.
- Knight F. H., (1921) "Risk, Uncertainty, and Profit", Boston, Houghton Mifflin.
- Kovner A., Vickery J., and Zhou L., (2014), "Do Big Banks Have Lower Operating Costs?", FRBNY, Economic Policy Review, Special Issue, March.
- International Capital Markets Associations, (ICMA), (2017), "Bail-in Workshop, Joining the Dots", Issue No. 2.
- Laeven L., (2011), "Banking Crises: A Review", Annual Review of Financial Economics, vol. 3.
- Leland H. E., and Pyle D. H., (1977), "Information Asymmetries, Financial Structure and Financial Intermediaries", Journal of Finance, Vol. 32, Issue No. 2.
- Lewrick U., Serena J. M., and Turner G., (2019), "Believing in bail-in? Market discipline and the pricing of bail-in bonds", BIS Working Papers, No. 831.
- Majnoni G., Bernardini G., Dal Santo A., and Trapanese M., (2021), "The EU bank insolvency framework: could less be more?", Bank of Italy, Occasional Papers No. 594, February.
- Martynova N., and Perotti E., (2015), "Convertible Bonds and Bank Risk-Taking", De Nederlandsche Bank, Working Paper No. 480.
- McAndrews J., Morgan D. P., Santos J. A. C., and Yorulmazer T., (2014), "What Makes Large Bank Failures So Messy and What Should be Done About It?", FRBNY Economic policy Review, December.
- Merton R. C., (1977), "An Analytical Derivation of the Cost of Deposit Insurance and Loan Guarantees", Journal of Bank and Finance, Vol. 1, Issue 3.
- Merton R. C., and Bodie Z., (1993), "Deposit Insurance Reform: A Functional Approach", Carnegie-Rochester Conference on Public Policy, Vol. 38, Issue 1.
- Micossi S., Bruzzone G., and Cassella M., (2014), "Bail-in Provisions in State Aid and Resolution Procedures: Are they consistent with systemic stability", Centre for European Policy Studies (CEPS), Policy Brief No. 318.
- Morgan D. P., and Stiroh K. J., (2005), "Too Big to Fail After All These Years", FRBNY Staff Report No 220.
- Morrison A. D., (2011), "Systemic risks and the `too-big-to-fail' problem", Oxford Review of Economic Policy, Vol. 27, Issue No. 3.

- Otker-Robe I., Narain A., Ilyina A., and Surti J., (2011), "The Too-important-toFail Conundrum: Impossible to Ignore and Difficult to Resolve", IMF, Staff Document Note, May.
- Pagano M., (2016), "Lessons from the European Financial Crisis', CFS Working Papers Series, No. 486.
- Pandolfi L., (2018), "Bail-in vs Bailout: a False Dilemma?", Centre for Studies in Economics and Finance (CSEF), Working Paper No. 499, May.
- Pennacchi G., (2010), "A Structural Model of Contingent Bank Capital", FRB of Cleveland Working Paper No. 10.
- Persaud A., (2014), "Why bail-in securities are fool's gold", Available at SSRN.
- Philippon T., and Salord A., (2017), "Bail-in and Bank Resolution in Europe", International Center for Monetary and Banking Studies, Special Report No. 4, March.
- Pisani-Ferri J., Sapir A., Veron N., and Wolf G. B., (2012), "What Kind of a European Banking Union', Bruegel Policy Contribution, Issue No. 12.
- Repullo R., (2005), "Liquidity, Risk Taking, and the Lender of Last Resort", International Journal of Central Banking, Vol. 1.
- Ringe W. G., (2016), "Bail-in between Liquidity and Solvency", University of Oxford, Legal Research Paper Series, Paper No. 33.
- Roll E., (1977), "Storia del pensiero economico", Universale Scientifica Boringhieri, Torino.
- Romer C. D., (1993), "The Nation in Depression", Journal of Economic Perspectives, Vol. 7.
- Schafer A., Schnabel I., and Weder di Mauro B., (2016), "Bail-in Expectations for European Banks: Actions Speak Louder than Words", CEPR Discussion Paper No. 11061..
- Schleifer A., and Vishny R., (2011), "Fire Sales in Finance and Macroeconomics', Journal of Economic Perspectives, Vol. 25.
- Schneider, M., and Shields T. W., (2022), "Motives for cooperation in the one-shot prisoner's dilemma", Journal of Behavioral Finance, Vol. 23, Issue 4.
- Schumpeter J. A., (1979), "Storia dell'analisi economica", Universale Scientifica Boringhieri, Torino.
- Schweikhard F. A., and Tsesmelidakis Z., (2012), "The Impact of Government Interventions on CDS and Equity Markets", American Finance Association, Chicago Meetings Working Paper, available at SSRN.
- Sommer J. H., (2014), "Why Bail-in? And How?", FRBNY, Economic Policy Review, December.
- Stanford Encyclopedia of Philosophy, SEP, (2024), "Prisoner's Dilemma".

- Stein J., (2012), "Monetary Policy as Financial Stability Regulation", Quarterly Journal of Economics, 127, No. 1, February.
- Strahan P. E., (2013), "Too Big to Fail: Causes, Consequences, and Policy Responses", Annual Review of Financial Economics, 5.
- Tarullo D., (2013), "Toward Building A More Effective Resolution Regime: Progress and Challenges", Federal Reserve System, Board of Governors, Speeches, October.
- Trapanese M., (2022), "Regulatory complexity, uncertainty, and systemic risk", Bank of Italy, Occasional Papers Series, No. 698, June.
- Trapanese M, Albareto G., Cardillo S., Castagna M., Falconi R., Pezzullo G., Serafini L., and Signore F., (2024), "The 2023 US banking crises: causes, policy responses, and lessons", Bank of Italy, Occasional Papers series, No. 870, July.
- Troger T. H., (2015), "Regulatory Influence on Market Conditions in the Banking Union: the Cases of Macro-Prudential Instruments and the Bail-in Tool", European Business Organization Law Review, Vol. 16, Issue 575.
- Troger T. H., (2018), "Too Complex to Work: A Critical Assessment of the Bail-in Tool under the European Bank Recovery and Resolution Regime", Journal of Financial Regulation, Vol. 4.
- Tsesmelidakis Z., and Merton R. C., (2012), "The Value of Implicit Guarantees", available at http://ssrn.com/abstract1/42231317.
- Tucker P., (2013), "Resolution and the Future of Finance", The Hague, May.
- Ueda K., and Weder-Di Mauro B., (2012), "Quantifying the Value of the Subsidy for Systemically Important Financial Institutions", IMF Working Paper No 12/128.
- Ventoruzzo M., and Sandrelli G. (2020), "O Tell Me the Truth about Bail-in: Theory and Practice", Journal of Business, Entrepreneurship, & Law, Vol. 13, Issue 1.
- Wojcik K. P., (2016), "Bail-in in the Banking Union", Common Market Law Review, Vol. 53, Issue No. 91.
- Zhou J., Rutledge V., Bossu W., Dobler M., Jassaud N., and Moore M., (2012), "From Bail-out to Bail-in: Mandatory Debt Restructuring of Systemic Financial Institutions", IMF Staff Discussion Note, No. 12, April.